\documentclass[a4, 12pt]{amsart}
\usepackage{a4wide,amssymb}
\usepackage{yfonts}
\usepackage{hyperref}
\renewcommand{\subset}{\subseteq}
\def\pointir{\unskip . --- \ignorespaces}
\newtheorem{guess}{Conjecture}

\newtheorem*{remark*}{Remark}
\newtheorem{guesss}{Conjecture}[section]
\newtheorem{proposition}{Proposition}[section]
\newtheorem{corollary}{Corollary}[section]
\newtheorem{lemma}{Lemma}[section]
\newtheorem{remark}{Remark}
\newtheorem{theorem}{Theorem}[section]
\newtheorem*{theorem*}{Theorem}

\newcommand{\Hom}{\operatorname{Hom}}
\newcommand{\End}{\operatorname{End}}
\newcommand{\Aut}{\operatorname{Aut}}
\newcommand{\Iso}{\operatorname{Iso}}
\newcommand{\Det}{\operatorname{Det}}
\newcommand{\Ker}{\operatorname{Ker}}
\newcommand{\tr}{\operatorname{Tr}}
\newcommand{\GL}{\operatorname{GL}}
\newcommand{\ad}{\operatorname{ad}}
\newcommand{\Fr}{\operatorname{Fr}}
\newcommand{\Supp}{\operatorname{Supp}}
\newcommand{\Id}{\operatorname{Id}}
\renewcommand{\Im}{\operatorname{Im}}
%\newcommand{\dim}{{\operatorname{dim}}}



\makeatletter

\title{ Multiplicity One Theorems}
\author[Aizenbud]{Avraham Aizenbud}
\address{A. Aizenbud, Faculty of Mathematics and Computer Science, Weizmann
Institute of Science POB 26, Rehovot 76100, ISRAEL.}
\email{aizenr@yahoo.com}

\author[Gourevitch]{Dmitry Gourevitch}

\address{D. Gourevitch,
Faculty of Mathematics and Computer Science, Weizmann Institute of
Science POB 26, Rehovot 76100, ISRAEL.} \email{guredim@yahoo.com}
\thanks{The first-named and second-named authors are partially supported by GIF Grant 861/05 and ISF Grant 1438/06}

\author[Rallis]{Steve Rallis}
\address{S.Rallis, Department of Mathematics Ohio State
University, COLUMBUS, OH 43210}
\email{E-mail:haar@math.ohio-state.edu}
\thanks{The third-named author is partially supported by NSF Grant DMS-0500392}
\author[Schiffmann]{G\'erard Schiffmann}
\address{G.Schiffmann, Institut de Recherche Math\'ematique
Avanc\'ee, Universit\'e Louis-Pasteur et CNRS, 7, rue
Ren\'e-Descartes, 67084 Strasbourg Cedex, France.}
\email{schiffma@math.u-strasbg.fr}


\begin{document}
\begin{thebibliography}{GPSR97}

\bibitem[AG07]{AG}
A.~Aizenbud and D.~Gourevitch, \emph{A proof of the multiplicity
one conjecture
  for {$GL(n)$} in {$GL(n + 1)$}.}, arXiv:0707.2363v2 [math.RT] (2007).

\bibitem[AG08]{AG2}
\bysame, \emph{Multiplicity one theorem for {$({\rm
GL}_{n+1}(\mathbb{R}),{\rm
  GL}_n(\mathbb{R}))$}}, arXiv:0808.2729 [math.RT] (2008).

\bibitem[AGS08a]{AGS1}
A.~Aizenbud, D.~Gourevitch, and E.~Sayag, \emph{{$({\rm
GL}_{n+1}(F),{\rm GL}_n(F))$} is a {G}elfand pair for any local
field {$F$}}, arXiv:0709.1273v3 [math.RT] (2008).

\bibitem[AGS08b]{AGS2}
\bysame, \emph{{$(O(V + F),O(V ))$} is a
  {G}elfand pair for any quadratic space {$V$} over a local field {$F$}}, Math.
  Z. (2008), see also arXiv:0711.1471[math.RT].



\bibitem[Ber84]{Ber}
Joseph~N. Bernstein, \emph{{$P$}-invariant distributions on {${\rm
GL}(N)$} and
  the classification of unitary representations of {${\rm GL}(N)$}
  (non-{A}rchimedean case)}, Lie group representations, II (College Park, Md.,
  1982/1983), Lecture Notes in Math., vol. 1041, Springer, Berlin, 1984,
  pp.~50--102. \MR{MR748505 (86b:22028)}

\bibitem[BZ76]{BZ}
I.~N. Bern{\v{s}}te{\u\i}n and A.~V. Zelevinski{\u\i},
\emph{Representations of
  the group {$GL(n,F),$} where {$F$} is a local non-{A}rchimedean field},
  Uspehi Mat. Nauk \textbf{31} (1976), no.~3(189), 5--70. \MR{MR0425030 (54
  \#12988)}

\bibitem[GK75]{GK}
I.~M. Gel{\cprime}fand and D.~A. Kajdan, \emph{Representations of
the group
  {${\rm GL}(n,K)$} where {$K$} is a local field}, Lie groups and their
  representations (Proc. Summer School, Bolyai J\'anos Math. Soc., Budapest,
  1971), Halsted, New York, 1975, pp.~95--118. \MR{MR0404534 (53 \#8334)}

\bibitem[GP92]{GP}
Benedict~H. Gross and Dipendra Prasad, \emph{On the decomposition
of a
  representation of {${\rm SO}\sb n$} when restricted to {${\rm SO}\sb
  {n-1}$}}, Canad. J. Math. \textbf{44} (1992), no.~5, 974--1002. \MR{MR1186476
  (93j:22031)}

\bibitem[GPSR97]{GPSR}
D.~Ginzburg, I.~Piatetski-Shapiro, and S.~Rallis, \emph{{$L$}
functions for the
  orthogonal group}, Mem. Amer. Math. Soc. \textbf{128} (1997), no.~611,
  viii+218. \MR{MR1357823 (98m:11041)}

\bibitem[GR06]{GR}
Benedict~H. Gross and Mark Reeder, \emph{From Laplace to Langlands
via representations of orthogonal groups.}, Bull. Amer. Math. Soc.
(N.S.)
  \textbf{43} (2006), no.~2, 163--205. \MR{MR2216109 (2007a:11159)}

\bibitem[Hak03]{Hi}
Jeffrey Hakim, \emph{Supercuspidal gelfand pairs.}, J. Number
Theory
  \textbf{100} (2003), no.~2, 251--269. \MR{MR1978455 (2004b:22017)}

\bibitem[HC99]{H}
Harish-Chandra, \emph{Admissible invariant distributions on
reductive
  {$p$}-adic groups}, University Lecture Series, vol.~16, American Mathematical
  Society, Providence, RI, 1999, Preface and notes by Stephen DeBacker and Paul
  J. Sally, Jr. \MR{MR1702257 (2001b:22015)}

\bibitem[JR96]{JR}
Herv{\'e} Jacquet and Stephen Rallis, \emph{Uniqueness of linear
periods},
  Compositio Math. \textbf{102} (1996), no.~1, 65--123. \MR{MR1394521
  (97k:22025)}

\bibitem[Kat03]{KSM}
Atsushi; Sugano~Takashi Kato, Shin-ichi;~Murase,
\emph{Whittaker-shintani
  functions for orthogonal groups.}, Tohoku Math. J. \textbf{55} (2003), no.~2,
  1--64. \MR{MR1956080 (2003m:22020)}

\bibitem[Mur96]{MS}
Takashi Murase, Atsushi;~Sugano, \emph{Shintani functions and
automorphic
  $l$-functions for ${\rm gl}(n)$.}, Tohoku Math. J. \textbf{48} ((1996)),
  no.~(2), 165--202. \MR{MR1387815 (97i:11056)}

\bibitem[MVW87]{MVW}
Colette M{\oe}glin, Marie-France Vign{\'e}ras, and Jean-Loup
Waldspurger,
  \emph{Correspondances de {H}owe sur un corps {$p$}-adique}, Lecture Notes in
  Mathematics, vol. 1291, Springer-Verlag, Berlin, 1987. \MR{MR1041060
  (91f:11040)}

\bibitem[Pra90]{Prasad}
Dipendra Prasad, \emph{Trilinear forms for representations of
{${\rm GL}(2)$}
  and local {$\epsilon$}-factors}, Compositio Math. \textbf{75} (1990), no.~1,
  1--46. \MR{MR1059954 (91i:22023)}

\bibitem[Pra93]{P2}
\bysame, \emph{On the decomposition of a representation of {${\rm
GL}(3)$}
  restricted to {${\rm GL}(2)$} over a {$p$}-adic field}, Duke Math. J.
  \textbf{69} (1993), no.~1, 167--177. \MR{MR1201696 (93m:22019)}

\bibitem[RS77]{RS2} S.~Rallis and G.~Schiffmann,
\emph{Repr\'esentations supercuspidales du groupe
m\'etaplectique}, J. Math. Kyoto Univ. \textbf{17} (1977), no.~3,
567-603. \MR{MR0498395 (58 \#16523)}

\bibitem[RS07]{RS1}
S.~Rallis and G.~Schiffmann, \emph{Multiplicity one conjectures},
  arXiv:0705.2168v1 [math.RT] (2007).

\bibitem[SS70]{SS}
T.~A. Springer and R.~Steinberg, \emph{Conjugacy classes}, Seminar
on Algebraic
  Groups and Related Finite Groups (The Institute for Advanced Study,
  Princeton, N.J., 1968/69), Lecture Notes in Mathematics, Vol. 131, Springer,
  Berlin, 1970, pp.~167--266. \MR{MR0268192 (42 \#3091)}

\bibitem[SZ08]{SZ}
B.~Sun and C.-B. Zhu, \emph{Multiplicity one theorems: the
archimedean case},  preprint availiable at
$http://www.math.nus.edu.sg/\sim matzhucb/Multiplicity
\underline{\,\,\,}One.pdf$ (2008).


\bibitem[vD08]{vD}
G.~van Dijk, \emph{{$(U(p, q), U(p - 1, q))$} is a generalized
Gelfand pair},  preprint (2008).

\end{thebibliography}

%\large
%
%\def\section
%{\@startsection {section}{1}
%{-1pt}{-5ex \@plus -1ex \@minus -.2ex}
%{2ex \@plus .2ex}
%{\normalfont \large \bfseries } }

\maketitle


\begin{abstract}
In the local, characteristic 0, non-archimedean case,  we consider
distributions on $GL(n+1)$ which are invariant under conjugation
by $GL(n)$. We prove that such distributions are invariant
by transposition. This  implies multiplicity at most one for
restrictions from $GL(n+1)$ to $GL(n)$. \hfill\break Similar
Theorems are obtained for orthogonal or unitary
groups.\end{abstract}



\section*{Introduction} Let ${\mathbb F }$ be a non-archimedean
local field of characteristic 0. Let $W$ be a vector space over
${\mathbb F}$ of finite dimension $n+1\geqslant 1$ and let
$W=V\oplus U$ be a direct sum decomposition with $\dim V=n $. Then
we have an imbedding of $GL(V)$ into $GL(W)$. Our goal is to prove
the following Theorem:

%{\parindent 0pt
%{\bf Theorem $\mathbf 1$:}
\begin{theorem*}[1] \label{RepGL}
 If $\pi$ (resp. $\rho$) is an irreducible admissible representation of $GL(W)$ (resp. of $GL(V)$) then
\[ \dim \left (\Hom_ {GL(V)}(\pi _{|GL(V)}\, ,\, \rho )\right )\leqslant 1.\]
\end{theorem*}

We choose a basis of $V$ and a non-zero vector in $U$ thus getting a basis of $W$. We can identify $GL(W)$ with $GL(n+1,{\mathbb F})$ and $GL(V)$ with $GL(n,{\mathbb F})$. The transposition map is an involutive anti-automorphism of $GL(n+1,{\mathbb F)}$ which leaves $GL(n,{\mathbb F})$ stable. It acts on the space of distributions on $GL(n+1,{\mathbb F})$.

Theorem 1 is a Corollary of
\begin{theorem*}[2] \label{DistGL}
 A distribution on $GL(W)$
which is invariant under conjugation by $G=GL(V)$ is invariant by
transposition.
\end{theorem*}


One can raise a similar question for orthogonal and unitary
groups. Let ${\mathbb D}$ be   either ${\mathbb F}$ or a
quadratic extension of ${\mathbb F}$. If $x\in{\mathbb D}$ then
$\overline x$ is the conjugate of $x$ if ${\mathbb D}\neq{\mathbb
F}$ and is equal to $x$ if ${\mathbb D}={\mathbb F}$.

Let $W$ be a vector space over ${\mathbb D}$ of finite dimension $n+1\geqslant 1$.  Let $\langle  .,.\rangle  $ be a non-degenerate hermitian form on $W$. This form is bi-additive and
\[ \langle  dw,d'w'\rangle  =d\,\,\overline{d'}\langle  w,w'\rangle   ,\quad \langle  w',w\rangle  =\overline{\langle  w,w'\rangle  }.\]
Given a ${\mathbb D}$-linear map $u$ from $W$ into itself, its adjoint $u^*$ is defined by the usual formula
\[ \langle  u(w),w'\rangle  =\langle  w,u^*(w')\rangle . \]

Choose a vector $e$ in $W$ such that $\langle  e,e\rangle  \ne 0$; let $U={\mathbb D}e$ and $V=U^{\perp}$ the orthogonal complement. Then $V$ has dimension $n$ and the restriction of the hermitian form to $V$ is non-degenerate.

Let $M$ be the unitary group of $W$ that is to say the group of
all ${\mathbb D}$-linear maps $m$ of $W$ into itself which
preserve the hermitian form or equivalently such that $mm^*=1$.
Let $G$ be the unitary group of $V$. With the p-adic topology both
groups are of type lctd (locally compact, totally discontinuous)
and countable at infinity. They are reductive groups of classical
type.

The group $G$ is naturally imbedded into $M$.


\begin{theorem*}[1'] \label{RepO} If $\pi $ (resp $\rho $) is an irreducible
admissible representation of $M$ (resp of $G$) then
\[ \dim\left  (\Hom_G(\pi _{| G},\rho )\right  )\leq 1.\]
\end{theorem*}


Choose a basis $e_1,\dots e_n$ of $V$ such that $\langle
e_i,e_j\rangle  \in{\mathbb F}$. For
\[ w=x_0e+\sum _1^nx_ie_i\]
put
\[ \overline w=\overline x_0e+\sum _1^n\overline {x_i}\, e_i.\]
If $u$ is a ${\mathbb D}$-linear map from $W$ into itself, let $\overline u$ be defined by
\[ \overline u(w)=\overline{u(\overline w)}.\]


Let $\sigma $ be the anti-involution $\sigma (m)=\overline m^{-1}$
of $M$;  Theorem 1' is a consequence of
\begin{theorem*}[2'] \label{DistO}
A distribution on $M$ which is invariant under conjugation by $G$
is invariant under $\sigma $.
\end{theorem*}

\subsection*{The structure of our proof}

Let us describe briefly our proof. In section \ref{SecDist2Rep} we
recall why Theorem 2 (2') implies Theorem 1(1'). This idea goes to
back Gelfand-Kazhdan (\cite{GK}).

For the proofs of Theorems 2 and 2' we systematically use two
classical results : Bernstein's localization principle and a
variant of Frobenius reciprocity which we call Frobenius descent.
For the convenience of the reader they are both recalled in
section \ref{SecPrel}.

Then we proceed with $\GL(n)$. The proof is by induction on $n$;
the case $n=0$ is trivial. In general we first linearize the
problem by replacing the action of $\GL(V)$ on $\GL(W)$ by the
action on the Lie algebra of $\GL(W)$. As a $\GL(V)$-module this
Lie algebra is isomorphic to a direct sum $\mathfrak{g}\oplus
V\oplus V^*\oplus\mathbb{F}$ with $\mathfrak{g}$ the Lie algebra
of $G=\GL(V)$ and $V^*$ the dual space of $V$. The group $G$ acts
trivially on $\mathbb{F}$, by the adjoint action on its Lie
algebra and the natural actions on $V$ and $V^*$. The component
$\mathbb{F}$ plays no role. Let $u$ be a linear bijection of $V$
onto $V^*$ which transforms some basis of $V$ into its dual basis.
The involution may be taken as
$$(X,v,v^*)\mapsto (u^{-1}{\,}^tX\, u,u^{-1}(v^*),u(v)).$$
We have to show that a distribution $T$ on $\mathfrak{g}\oplus V\oplus V^*$ which is invariant under $G$ and skew relative to the involution is 0.

In section \ref{SecRedSingSet} we prove that the support of such a
distribution is contained in the set of singular elements. On the
$\mathfrak{g}$ side, using Harish-Chandra descent we get that the
support of $T$ must be contained in $(\mathfrak{z}+
\mathcal{N})\times (V\oplus V^*)$ where $\mathfrak{z}$ is the
center of $\mathfrak{g}$ and $\mathcal{N}$ the cone of nilpotent
elements in $\mathfrak{g}$. On the $V\oplus V^*$ side we show that
the support must be contained in $\mathfrak{g}\times\Gamma$ where
$\Gamma$ is the cone $\langle v, v^*\rangle = 0$ in $V\oplus V^*$.
On $\mathfrak{z}$ the action is trivial so we are reduced to the
case of a distribution on $\mathcal{N}\times\Gamma$.

In section \ref{SecEndGL} we consider such distributions. The end
of the proof is based on two facts. First, viewing the
distribution as a distribution on $\mathcal{N}\times (V\oplus
V^*)$ its partial Fourier transform relative to $V\oplus V^*$ has
the same invariance properties and hence must also be supported on
$\mathcal{N}\times \Gamma$. This implies in particular a
homogeneity condition on $V\oplus V^*$. The idea of using Fourier
transform in this kind of situation goes back at least to
Harish-Chandra (\cite{H}) and is conveniently expressed using a
particular case of the Weil or oscillator representation.

For $(v,v^*)\in\Gamma$, let $X_{v,v^*}$ be the map $x\mapsto \langle x,v^*\rangle v$ of $V$ into itself. The second fact is that the one parameter group of transformations
$$(X,v,v^*)\mapsto (X+\lambda X_{v,v^*},v,v^*)$$
is a group of (non-linear) homeomorphisms of
$[\mathfrak{g},\mathfrak{g}]\times\Gamma$ which commute with $G$
and the involution. It follows that the image of the support of
our distribution must also be singular. This allows us to replace
the condition $\langle v,v^*\rangle =0$ by the stricter condition
$X_{v,v^*}\in \Im \,\,\ad X$.

Using the stratification of $\mathcal{N}$ we proceed one nilpotent
orbit at a time, transferring the problem to $V\oplus V^*$ and a
fixed nilpotent matrix $X$. The support condition turns out to be
compatible with direct sum so that it is enough to consider the
case of a principal nilpotent element. In this last situation the
key is the homogeneity condition coupled with an easy induction.

The orthogonal and unitary cases are proved in a similar vein. In
section \ref{SecRedSingSetO} we reduce the support to the singular
set. Here the main difference is that we use Harish-Chandra
descent directly on the group. Note that the Levi subgroups have
components of type $\GL$ so that Theorem 2 has to be used. Finally
in section \ref{SecEndO} we consider the case of a distribution
whose support is contained in the set of singular elements; the
proof is along the same lines as in section \ref{SecEndGL}.

\subsection*{Remarks}

%Similar theorems should be true in the archimedean case.
As for the archimedean case, partial analogs of the results of
this paper were obtained in \cite{AGS1, AGS2, vD}. Recently, the
full analogs were obtained in \cite{AG2} and \cite{SZ}.

Let us add some comments on the Theorems themselves. First note
that Theorem 2 gives an independent proof of a well known theorem
of Bernstein: choose a basis $e_1,\dots ,e_n$ of $V$, add some
vector $e_0$ of $W$ to obtain a basis of $W$ and let $P$ be the
isotropy of $e_0$ in $\GL(W)$. Then Theorem B of \cite{Ber} says
that a distribution on $\GL(W)$ which is invariant under the
action of $P$ is invariant under the action of $\GL(W)$. Now, by
Theorem 2 such a distribution is invariant under conjugation by
the transpose of $P$ and the
 group of inner automorphisms is generated by the images of $P$ and its transpose.
This Theorem implies Kirillov's conjecture which states that any
unitary irreducible representation of $\GL(W)$ remains irreducible
when restricted to $P$.

The occurrence of involutions in multiplicity  at most one
problems is of course nothing new. The situation is fairly simple
when all the orbits are stable by the involution thanks to
Bernstein's localization principle and constructivity theorem
(\cite{BZ, GK}). In our case this is not true : only generic
orbits are stable. Non-stable orbits may carry invariant measures
but they do not extend to the ambient space (a similar situation
is already present in \cite{Ber}).

 An illustrative example is the case $n=1$ for $\GL$.
It reduces to $\mathbb{F}^*$ acting on $\mathbb{F}^2$ as
$(x,y)\mapsto (tx,t^{-1}y)$. On the $x$ axis the measure
$d^*x=dx/|x|$ is invariant but does not extend invariantly.
However the symmetric measure
$$f\mapsto \int_{\mathbb{F}^*}f(x,0)d^*x+\int_{\mathbb{F}^*}f(0,y)d^*y$$
does extend.

As in similar cases (for example \cite{JR}) our proof does not
give a simple explanation of why all invariant distributions are
symmetric. The situation would be much better if we had some kind
of density theorem. For example in the $\GL$ case let us say that
an element $(X,v,v^*)$ of $\mathfrak{g}\oplus V\oplus V^*$ is
regular if $(v,Xv,\dots X^{n-1}v)$ is a basis of $V$ and
$(v^*,\dots ,{}^tX^{n-1}v^*)$ is basis of $V^*$. The set of
regular elements is a non-empty Zariski open subset; regular
elements have trivial isotropy subgroups. The regular orbits are
the orbits of the regular elements; they are closed, separated by
the invariant polynomials and stable by the involution (see
\cite{RS1}). In particular they carry invariant measures which,
the orbits being closed, do extend and are invariant by the
involution. It is tempting to conjecture that the subspace of the
space of invariant distributions generated by these measures is
weakly dense. This would provide a better understanding of Theorem
2. Unfortunately if true at all, such a density theorem is likely
to be much harder to prove.

Assuming multiplicity at most one, a more difficult question is to find when it is one. Some partial results are known.

For the orthogonal group (in fact the special orthogonal group)
this question has been studied by B.~Gross and D.~Prasad
(\cite{GP, P2}) who formulated a precise conjecture. An up to date
account is given by B.~Gross and M.~Reeder (\cite{GR}). In a
different setup, in their work on "Shintani" functions A.~Murase
and T.~Sugano obtained complete results for $GL(n)$ and the split
orthogonal case but only for spherical representations (\cite{KSM,
MS}). Finally we should mention, Hakim's publication \cite{Hi},
which, at least for the discrete series, could perhaps lead to a
different kind of proof.

Multiplicity one theorems have important applications to
the relative trace formula, to automorphic descent,  to local and
global liftings of automorphic representations, and to
determinations of L-functions. In particular, multiplicity at most
one is used as a hypothesis in the work \cite{GPSR} on the study
of automorphic L-functions on classical groups. At least for the
last two authors, the original motivation for this work came in
fact from \cite{GPSR}.



\subsection*{Acknowledgements}
The first two authors would like to thank their teacher {\bf
Joseph Bernstein} for teaching them most of the mathematics they
know. They cordially thank {\bf Joseph Bernstein} and {\bf Eitan
Sayag} for guiding them through this project. They would also like
to thank {\bf Vladimir Berkovich}, {\bf Yuval Flicker}, {\bf Erez
Lapid}, {\bf Omer Offen} and {\bf Yiannis Sakellaridis} for useful
remarks.

The first two authors worked on this project  while participating
in the program \emph{Representation theory, complex analysis and
integral geometry} of the Hausdorff Institute of Mathematics (HIM)
at Bonn joint with Max Planck Institute fur Mathematik. They wish
to thank the organizers of the activity and the director of HIM
for inspiring environment and perfect working conditions.

Finally, the first two authors wish to thank {\bf Vladimir
Berkovich}, {\bf Stephen Gelbart}, {\bf Maria Gorelik} and {\bf
Sergei Yakovenko} from the Weizmann Institute of Science for their
encouragement, guidance and support.

The last author thanks the Math Research Institute of Ohio State University in Columbus for several invitations which allowed him to work with the third author.
% \hfill$\Box$\hfill
%
\section{Theorem 2(2') implies Theorem 1(1')}
\label{SecDist2Rep}

A group of type lctd is  a locally compact, totally disconnected
group which is countable at infinity. We consider smooth
representations of such groups. If $(\pi ,E_\pi )$ is such a
representation then $(\pi ^*,E_\pi ^*)$ is the smooth
contragradient. Smooth induction is denoted by $Ind$ and compact
induction by $ind$. For any topological space $T$ of type lctd,
$\mathcal {S}(T)$ is the space of functions locally constant,
complex valued, defined on $T$ and with compact support. The space
$\mathcal {S'}(T)$ of distributions on $T$ is the dual space to
$\mathcal {S}(T)$.
\begin{proposition} \label{GK_Crit}
 Let $M$ be a lctd group and $N$ a closed subgroup, both unimodular. Suppose that there exists an involutive anti-automorphism $\sigma $ of $M$ such that $\sigma (N)=N$ and such that any distribution on $M$, biinvariant under $N$, is fixed by $\sigma $. Then, for any irreducible admissible representation $\pi $ of $M$
\[ \dim \left  (\Hom _ M(ind_ N^M(1),\pi )\right  )\times \dim \left  (\Hom _ M(ind_ N^M(1),\pi ^*)\right  )\leq 1.\]
\end{proposition}
This is well known (see for example \cite{Prasad}).


\begin{remark*} There is a variant for the non-unimodular case;
we will not need it.
\end{remark*}

\begin{corollary} \label{GK_Cor}
 Let  $M$ be a lctd group and $N$ a closed subgroup, both unimodular. Suppose that there exists an involutive anti-automorphism $\sigma $ of $M$ such that $\sigma (N)=N$ and such that any distribution on $M$, invariant under conjugation by  $N$, is fixed by $\sigma $. Then, for any irreducible admissible representation $\pi $ of $M$ and any irreducible admissible representation $\rho $ of $N$
\[ \dim\left  (\Hom_ N(\pi _{|N},\rho ^*)\right  )\times \dim\left  (\Hom_ N((\pi ^*)_{|N},\rho )\right  )\leq 1.\]
\end{corollary}
\begin{proof} Let $M'=M\times N$ and $N'$ be the closed subgroup of
$M'$ which is the image of the diagonal embedding of $N$ in $M'$.
The map $(m,n)\mapsto mn^{-1}$ of $M'$ onto $M$ defines a
homeomorphism of $M'/N'$ onto $M$. The inverse map is $m\mapsto
(m,1)N'$. On $M'/N'$ left translations by $N'$ correspond to the
action of $N$ on $M$ by conjugation. We have a bijection between
the space of distributions $T$ on $M$ invariant under the action
of $N$ by conjugation and the space of distributions $S$ on $M'$
which are biinvariant under $N'$. Explicitly
\[ \langle  S,f(m,n)\rangle  =\langle  T,\int _Nf(mn,n)dn\rangle . \]
Suppose that $T$ is invariant under $\sigma $ and consider the involutive anti-automorphism $\sigma '$ of $M'$ given by $\sigma '(m,n)=(\sigma (m),\sigma (n))$. Then
\[ \langle  S,f\circ \sigma '\rangle  =\langle  T,\int _Nf(\sigma (n)\sigma (m),\sigma (n))dn\rangle.  \]
Using the invariance under $\sigma $ and for the conjugation action of $N$ we get
 \begin{eqnarray*}
 \langle  T,\int _Nf(\sigma (n)\sigma (m),\sigma (n))dn\rangle  &=&\langle  T,\int _Nf(\sigma (n)m,\sigma (n))dn\rangle  \\
&=&\langle  T,\int _Nf(mn,n)dn\rangle  \\
&=&\langle  S,f\rangle .
\end{eqnarray*}
\overfullrule=0pt Hence $S$ is invariant under $\sigma '$.
Conversely if $S$ is invariant under $\sigma '$ the same
computation shows that $T$ is invariant under $\sigma $. Under the
assumption of the corollary we can now apply Proposition
\ref{GK_Crit} and we obtain the inequality
 \[ \dim \left  (\Hom _ {M'}(ind_ {N'}^{M'}(1),\pi \otimes \rho )\right  )\times \dim \left  (\Hom _ {M'}(ind_ {N'}^{M'}(1),\pi ^*\otimes \rho ^*)\right  )\leq 1.\]
 We know that $Ind_ {N'}^{M'}(1)$ is the smooth contragredient representation of $ind_{N'}^{M'}(1)$; hence
 \[ \Hom _ {M'}(ind_ {N'}^{M'}(1),\pi ^*\otimes \rho ^*)\approx \Hom _ {M'}(\pi \otimes \rho , Ind _ {N'}^{M'}(1)).\]
 Frobenius reciprocity tells us that
 \[ \Hom _ {M'}\bigl (\pi \otimes \rho , Ind _ {N'}^{M'}(1)\bigr )\approx \Hom _ {N'}\bigl ((\pi \otimes \rho )_{|N'},1\bigr ).\]
Clearly
\[ \Hom _ {N'}\bigl ((\pi \otimes \rho )_{|N'},1\bigr )\approx \Hom _ {N}\bigl (\rho ,(\pi _ {|N})^*\bigr )\approx \Hom_ N(\pi _{|N},\rho ^*).\]
Using again Frobenius reciprocity we get
\[ \Hom _ {N}\bigl (\rho ,(\pi _ {|N})^*\bigr )\approx \Hom _ {M}\bigl (ind_ N^M(\rho ),\pi ^*\bigr ).\]
In the above computations we may replace $\rho $ by $\rho ^*$ and $\pi $ by $\pi ^*$. Finally
\begin{eqnarray*}
\Hom _ {M'}(ind_ {N'}^{M'}(1),\pi ^*\otimes \rho ^*)&\approx& \Hom_ {N}(\rho ,(\pi _ {|N})^*)\\
&\approx& \Hom_ N(\pi _{|N},\rho ^*)\\
&\approx& \Hom _ {M}(ind_ N^M(\rho ),\pi ^*).\\
\Hom _ {M'}(ind_ {N'}^{M'}(1),\pi \otimes \rho )
&\approx& \Hom _ {N}(\rho ^*,((\pi ^*)_ {|N})^*)\\
&\approx& \Hom_ N((\pi ^*)_{|N},\rho )\\
&\approx& \Hom _ {M}(ind_ N^M(\rho ^*),\pi ).
\end{eqnarray*}
\end{proof}


Consider the case $M=GL(W)$ and $N=GL(V)$ in the notation of the
intrduction. In order to use Corollary \ref{GK_Cor} to infer
Theorem 1 from Theorem 2 it remains to show that
\begin{equation} \label{homeq}
\Hom_ N((\pi ^*)_{|N},\rho )\approx \Hom_ N(\pi _{|N},\rho ^*)
\end{equation}
Let $E_\pi $ be the space of the representation $\pi$  and let
$E_\pi ^*$ be the smooth  dual (relative to the action of
$GL(W))$. Let $E_\rho $ be the space of $\rho $ and $E^*_\rho $ be
the smooth dual for the action of $GL(V)$. We know, \cite[section
7]{BZ}  that the contragredient representation $\pi ^*$ in $E_\pi
^*$ is isomorphic to the representation $g\mapsto \pi
(^tg^{{-1}})$ in $E_\pi $. The same is true for $\rho ^*$.
Therefore an element of $\Hom_ N(\pi _{|N},\rho ^*)$ may be
described as a linear map $A$ from $E_\pi $ into $E_\rho $ such
that, for $g\in N$
\[ A\pi (g)=\rho (^tg^{-1})A.\]
An element of $\Hom_ N((\pi ^*)_{|N},\rho )$ may be described as a
linear map $A'$ from $E_\pi $ into $E_\rho $ such that, for $g\in
N$
\[ A'\pi (^tg^{-1})=\rho (g)A'.\]
This yields (\ref{homeq}).


Similarly, we prove that Theorem 2' implies Theorem 1'. With the
notation of the introduction, this would follow from Corollary
\ref{GK_Cor} provided that
\begin{equation} \label{homeq2}
\Hom \left  (\pi ^*_{|G},\rho \right  )\approx \Hom\left (\pi
_{|G},\rho ^*\right  ).
\end{equation}
To show (\ref{homeq2}) we use the following result of \cite[Chapter 4]{MVW}.
Choose $\delta \in GL_{\mathbb F}(W)$ such that $\langle  \delta w,\delta w'\rangle  =\langle  w',w\rangle  $. If $\pi $ is an irreducible admissible representation of $M$, let $\pi ^*$ be its smooth contragredient and define $\pi ^\delta $ by
\[ \pi ^\delta (x)=\pi (\delta x\delta ^{-1}).\]
Then $\pi ^\delta $ and $\pi ^*$ are equivalent. We choose $\delta =1$ in the orthogonal case ${\mathbb D}={\mathbb F}$. In the unitary case, fix an orthogonal basis of $W$, say $e_1,\dots ,e_{n+1}$, such that $e_2,\dots ,e_{n+1}$ is a basis of $V$; put $\langle  e_i,e_i\rangle  =a_i$. Then
\[ \langle  \sum x_ie_i,\sum y_je_j\rangle  =\sum a_ix_i\overline{y_i}.\]
Define $\delta $ by
\[ \delta \left  (\sum x_ie_i\right  )=\sum \overline{x_i}e_i.\]
Note that $\delta ^2=1$.

Let $E_\pi $ be the space of $\pi $. Then, up to equivalence, $\pi
^*$ is the representation $m\mapsto \pi (\delta m\delta ^{-1})$.
If $\rho $ is an admissible irreducible representation of $G$ in a
vector space $E_\rho $ then an element $A$ of $\Hom\left (\pi
^*_{|G},\rho \right  )$ is a linear map from $E_\pi $ into $E_\rho
$ such that
\[ A\pi (\delta g\delta ^{-1})=\pi (g)A,\quad g\in G.\]
In turn the contragredient $\rho ^*$ of $\rho $  is equivalent to
the representation $g\mapsto \rho (\delta g\delta ^{-1})$ in
$E_\rho $. Then an element $B$ of $\Hom\left  (\pi _{|G},\rho
^*\right  )$ is a linear map from $E_\pi $ into $E_\rho $ such
that
\[ B\pi (g)=\rho (\delta g\delta ^{-1})B,\quad g\in G.\]
As $\delta ^2=1$ the conditions on $A$ and $B$ are the same.
Thus (\ref{homeq2}) follows.

From now on we concentrate on Theorems 2 and 2'.

\section{Some tools} \label{SecPrel}
We shall state two theorems which are systematically used in our
proof.

If $X$ is a Hausdorff totally disconnected  locally compact
topological space (lctd space in short) we denote by ${\mathcal
S}(X)$ the vector space of locally constant functions with compact
support of $X$ into the field of complex numbers ${\mathbb C}$.
The dual space ${\mathcal S}'(X)$ of ${\mathcal S}(X)$ is the
space of distributions on $X$ with the weak topology. All the lctd spaces we introduce
are countable at infinity.

If a lctd topological group  $G$ acts continuously on a lctd
space $X$ then it acts on ${{\mathcal S}}(X)$ by \[
(gf)(x)=f(g^{-1}x)\] and on distributions by
\[ (gT)(f)=T(g^{-1}f)\]
The space of invariant distributions is denoted by ${\mathcal
S}'(X)^G$. More generally, if $\chi$ is a character of $G$ we
denote by ${\mathcal S}'(X)^{G,\chi}$ the space of distributions
$T$ which transform according to $\chi$ that is to say
$gT=\chi(g)T$.


The following result is due to Bernstein \cite{Ber}, section 1.4.
\begin{theorem}[Localization principle] \label{ThmLocPrin}
Let  $q:Z \to T$ be a continuous map between two topological
spaces of type lctd. Denote $Z_t:= q^{-1}(t)$. Consider
$\mathcal{S}'(Z)$ as $\mathcal{S}(T)$-module. Let $M$ be a closed
subspace of $\mathcal{S}'(Z)$ which is an
$\mathcal{S}(T)$-submodule. Then $M=\overline{\bigoplus_{t \in T}
(M \cap \mathcal{S}'(Z_t))}$.
\end{theorem}

\begin{corollary} \label{LocPrin}
Let $q:Z \to T$ be a continuous map between topological spaces of
type lctd. Let a lctd group $H$ act on  $Z$ preserving the fibers
of $q$. Let $\mu$ be a character of $H$. Suppose that for any
$t\in T$, $\mathcal{S}'(q^{-1}(t))^{H,\mu}=0$. Then
$\mathcal{S}'(Z)^{H,\mu}=0$.
\end{corollary}

The second theorem is a variant of Frobenius reciprocity
(\cite[section 1.5]{Ber} and \cite[sections 2.21-2.36]{BZ} ).

\begin{theorem}[Frobenius descent] \label{Frob}
Suppose that a unimodular lctd topological group $H$ act
transitively on a lctd topological space $Z$. Let $\varphi:E \to
Z$ be an $H$-equivariant map of lctd topological spaces. Let $x\in
Z$. Assume that the stabilizer $Stab_H(x)$ is unimodular. Let
$W=\varphi^{-1}(x)$ be the fiber of $x$.
Let $\chi$ be a character
of $H$. Then
%
\begin{enumerate}
\item There exists a canonical isomorphism $\Fr:
\mathcal{S}'(E)^{H,\chi} \to \mathcal{S}'(W)^{Stab_H(x),\chi}$
given by
$$\langle \Fr(\xi), f \rangle = \int_Z \chi(g_z) \langle \xi,g_zf
\rangle dz,$$ where $dz$ denotes the Haar measure on $Z$, and $g_z
\in H$ is an element such that $g_z z =x$.
%
\item For any distribution $\xi \in \mathcal{S}'(E)^{H,\chi}$,
$\Supp(\Fr (\xi))=\Supp(\xi)\cap W$.
\end{enumerate}
\end{theorem}

In particular, consider the case where $H$ acts transitively on
$Z$ and $W$ is a finite dimensional vector space over $\mathbb F$
with a nondegenerate bilinear form $B$. Assume that $H$ acts on
$W$ linearly preserving $B$. Let $\Fr:\mathcal{S}'(Z\times
W)^{H,\chi}\rightarrow \mathcal{S}'(W)^{Stab_H(x)}$ be the
Frobenius isomorphism with respect to the projection map $Z\times
W\rightarrow Z$. Let $\mathcal{F}_B$ be the Fourier transform in
the $W$-coordinate. We have

\begin{proposition} \label{FrFouCommute}
For any $\xi \in \mathcal{S}'(Z\times W)^{H,\chi}$, we have
$\mathcal{F}_{B}(\Fr(\xi))=\Fr(\mathcal{F}_{B}(\xi))$
\end{proposition}

This Proposition will be used in sections \ref{SecEndGL} and
\ref{SecEndO}.

Finally as $\mathbb{F}$ is non-archimedean, a distribution which is 0 on some open set may be identified with a distribution on the (closed) complement.
This will be used throughout this work.
%
\section{Reduction to the singular set : the  GL(n)
case} \label{SecRedSingSet}
%If $ H$  is a topological group of type lctd, acting continuously
%on a topological space $ E$  of the same type and if $\chi $  is a
%continuous character of $ H$  we denote by $ {\mathcal
%S}'(E)^{H,\chi }$  the space of distributions $ T$  on $ E$  such
%that $ \langle T,f(h^{-1}x)\rangle =\chi (h)\langle T,f\rangle $  for
%any $ f\in  {\mathcal  S}(E)$  and any $ h\in  H$.
Consider the case of the general linear group. From the decomposition $ W=V\oplus {\mathbb   F}e$  we get, with obvious identifications
$$ \End  (W)=\End  (V)\oplus V\oplus V^*\oplus {\mathbb   F}.$$
Note that $ \End  (V)$  is the Lie algebra $ {\mathfrak g}$  of $
G$. The group $ G$  acts on $ \End  (W)$  by $
g(X,v,v^*,t)=(gXg^{-1},gv,{}^t g^{-1}v^*,t)$. As before choose a
basis $ (e_1,\dots ,e_n)$  of $ V$  and let $ (e_1^*,\dots
,e_n^*)$  be the dual basis of $ V^*$. Define an isomorphism $ u$
of $ V$  onto $ V^*$  by $ u(e_i)=e_i^*$. On $ GL(W)$  the
involution $ \sigma $  is $ h\mapsto u^{-1}{}^th^{-1}u$. It
depends upon the choice of the basis but the action on the space
of invariant distributions does not depend upon this choice.

 It will be convenient to introduce an extension $ \widetilde G$ of $G$
of degree two. Let $ \Iso (V,V^*)$  be the set of isomorphisms of
$ V$  onto $ V^*$. We define $ \widetilde  G=G\cup \Iso(V,V^*)$.
The group law, for $ g,g'\in  G$  and $ u,u'\in \Iso(V,V^*)$ is
$$ g \times g'=gg',\,\,\, u \times g=ug,\,\,\, g \times u=^t\hskip -3 pt g^{-1}u,\,\,\, u \times u'=^t\hskip -3 pt u^{-1}u'.$$
Now from $ W=V\oplus {\mathbb   F}e$  we obtain an identification
of the dual space $ W^*$  with $ V^*\oplus {\mathbb   F}e^*$  with
$ \langle e^*,V\rangle =(0)$  and $ \langle e^*,e\rangle =1$. Any
$ u$  as above extends to an isomorphism of $ W$  onto $ W^*$  by
defining $ u(e)=e^*$. The group $ \widetilde  G$  acts on $ GL(W)$
:
$$ h\mapsto ghg^{-1},\,\,\,\, h\mapsto ^t\hskip -3pt (uhu^{-1}),\quad g\in  G,\, h\in  GL(W),\,\, u\in   \Iso(V,V^*)$$
and also on $ \End  (W)$  with the same formulas.

Let $\chi $  be the character of $ \widetilde  G$  which is 1 on $
G$  and $ -1$  on $  \Iso(V,V^*)$. Our goal is to prove that $
{\mathcal  S}'(GL(W))^{\widetilde  G, \chi  }=(0)$.
\begin{proposition}
If $ {\mathcal  S}'({\mathfrak  g}\oplus V\oplus V^*)^{\widetilde
G, \chi  }=(0)$  then $ {\mathcal  S}'(GL(W))^{\widetilde  G, \chi
}=(0)$.
\end{proposition}
\begin{proof} We have $ \End(W)=\bigl (\,\,\End
(V)\oplus V\oplus V^*\bigr )\oplus {\mathbb   F}$
 and the action of $ \widetilde  G$  on $ {\mathbb   F}$  is
 trivial.
 Thus  $ {\mathcal  S}'({\mathfrak  g}\oplus V\oplus V^*)^{\widetilde  G, \chi  }=(0)$  implies that $ {\mathcal  S}'(\End   (W))^{\widetilde  G, \chi  }=(0).$
Let $ T\in  {\mathcal  S}'(GL(W))^{\widetilde  G, \chi  }$. Let $
h\in  GL(W)$  and choose a compact open neighborhood $ K$  of $
\Det \,\,  h$  such that $ 0\notin K$. For $ x\in \End (W)$ define
$ \varphi (x)=1$  if $  \Det  x\in  K $ and $ \varphi (x)=0$
otherwise. Then $ \varphi $  is a locally constant function. The
distribution $ (\varphi _{|GL(W)})T$  has a support which is
closed in $  \End (W)$  hence may be viewed as a distribution on $
\End (W)$. This distribution belongs to $ {\mathcal S}'(\End
(W))^{\widetilde  G, \chi  }$  so it must be equal to 0. It
follows that $ T$  is 0 in the neighborhood of $ h$. As $ h$ is
arbitrary we conclude that $ T=0$. \end{proof}

Our task is now to prove that $ {\mathcal  S}'({\mathfrak g}\oplus
V\oplus V^*)^{\widetilde  G, \chi  }=(0)$. We shall use induction
on the dimension $ n$  of $ V$.  The action of $ \widetilde  G$
is, for $ X\in  {\mathfrak  g},\, v\in  V,\, v^*\in  V^*,\, g\in
G,\,  u\in   \Iso(V,V^*)$
$$ (X,v,v^*)\mapsto (gXg^{-1},gv,^t\hskip -3pt g^{-1}v^*),\,\,\, (X,v,v^*)\mapsto (^t\hskip -1pt (uXu^{-1}),^t\hskip -3pt u^{-1}v^*,uv).$$
The case $n=0$  is trivial.



We suppose that  $ V$   is of dimension $ n\geq 1$, assuming the
result up to dimension $ n-1$ and for all $\mathbb F$. If $ T\in
{\mathcal  S}'({\mathfrak  g}\oplus V\oplus V^*)^{\widetilde  G,
\chi  }$  we are going to show that its support   is contained in
the "singular set". This will be done in two stages.

On $ V\oplus V^*$  let $ \Gamma $  be the cone $ \langle
v^*,v\rangle =0$. It is stable under $ \widetilde  G$.
\begin{lemma} \label{SuppTinGamma}
 The support of $ T$  is contained in $ {\mathfrak  g} \times \Gamma $.
\end{lemma}

\begin{proof} For $ (X,v,v^*)\in  {\mathfrak  g}\oplus V\oplus V^*$
put $ q(X,v,v^*)=\langle v^*,v\rangle $. Let $ \Omega $  be the
open subset $ q\ne 0$. We have to show that $ {\mathcal S}'(\Omega
)^{\widetilde  G, \chi  }=(0)$. By Bernstein's localization
principle (Corollary \ref{LocPrin}) it is enough to prove that,
for any fiber $ \Omega _t=q^{-1}(t),\,\,\, t\ne 0$, one has $
{\mathcal S}'(\Omega _t)^{\widetilde  G, \chi  }=(0)$.

 $ G$  acts transitively on the quadric $ \langle v^*,v\rangle =t$. Fix a decomposition $ V={\mathbb   F}\varepsilon \oplus V_1$  and identify $ V^*={\mathbb   F}\varepsilon ^*\oplus V_1^*$  with $ \langle \varepsilon ^*,\varepsilon \rangle =1$. Then $ (X,\varepsilon ,t\varepsilon ^*)\in  \Omega _t$
 and the isotropy subgroup of $ (\varepsilon ,t\varepsilon ^*)$  in $ \widetilde  G$  is, with an obvious notation $ \widetilde  G_{n-1}$.
 By Frobenius descent (Theorem \ref{Frob}) there is a linear bijection between $ {\mathcal  S}'(\Omega _t)^{\widetilde  G, \chi  }$  and the space $ {\mathcal  S}'({\mathfrak  g})^{\widetilde  G_1, \chi  _1}$  and this last space is $ (0)$  by induction.
\end{proof}


Let $ {\mathfrak  z}$  be the center of $ {\mathfrak  g}$  that is
to say the space of scalar matrices. Let $ {\mathcal  N}\subset
[{\mathfrak  g},{\mathfrak  g}]$  be the nilpotent cone in $
{\mathfrak  g}$.

\begin{proposition}
If $ T\in  {\mathcal  S}'({\mathfrak  g}\oplus V\oplus V^*)^{\widetilde  G, \chi  }$ then the support of $T$ is contained in $({\mathfrak z}+{\mathcal N})\times{\Gamma}$.\hfill\break
 If  $ {\mathcal  S}'({\mathcal  N} \times\Gamma)^{\widetilde  G, \chi  }=(0)$  then  $ {\mathcal  S}'({\mathfrak  g}\oplus V\oplus V^*)^{\widetilde  G, \chi  }=(0)$.
\end{proposition}
\begin{proof} Let us prove that the support of such a distribution
$T$ is contained in $({\mathfrak z}+{\mathcal N})\times (V\oplus
V^*)$. We use Harish-Chandra's descent method.  For $ X\in
{\mathfrak g}$  let $ X=X_s+X_n$  be the Jordan decomposition of $
X$  with $ X_s$  semisimple and $ X_n$  nilpotent. This
decomposition commutes with the action of $ \widetilde  G$. The
centralizer $ Z_G(X)$  of an element $ X\in  {\mathfrak  g}$   is
unimodular (\cite[page 235]{SS}) and there exists an isomorphism $
u$  of $ V$ onto $ V^*$  such that $ ^tX=uXu^{-1}$  (any matrix is
conjugate to its transpose). It follows that the centralizer $
Z_{\widetilde G}(X)$  of $ X$  in $ \widetilde  G$, a semi direct
product of $ Z_G(X)$  and $ S_2$, is also unimodular.

Let $E$  be the space of monic polynomials of degree $ n$ with
coefficients in $ {\mathbb   F}$. For $ p\in  E$, let $ {\mathfrak
g}_p$  be the set of all $ X\in  {\mathfrak  g}$  with
characteristic polynomial $ p$. Note that $ {\mathfrak  g}_p$  is
fixed by $ \widetilde  G$. By Bernstein localization principle
(Theorem \ref{ThmLocPrin}) it is enough to prove that if $ p$  is
not $ (T-\lambda )^n$  for some $ \lambda $  then $ {\mathcal
S}'({\mathfrak  g}_p \times V \times V^*)^{\widetilde  G, \chi
}=(0)$.

Fix $ p$. We claim that the map $ X\mapsto X_s$  restricted to $
{\mathfrak  g}_p$  is continuous. Indeed let $ \widetilde {\mathbb
F}$  be a finite Galois extension of $ {\mathbb   F}$ containing
all the roots of $ p$. Let
$$ p(\xi )=\prod _1^s(\xi -\lambda_i)^{n_i}$$
be the decomposition of $ p$. Recall that if $ X\in  {\mathfrak
g}_p$  and $ V_i=\Ker   (X-\lambda _i)^{n_i}$  then $ V=\oplus
V_i$  and the restriction of $ X_s$  to $ V_i$  is the
multiplication by $ \lambda _i$. Then choose a polynomial $ R$,
with coefficients in $ \widetilde  {\mathbb   F}$  such that for
all $ i$, $ R$  is congruent to $ \lambda _i$  modulo $ (\xi
-\lambda _i)^{n_i}$  and $ R(0)=0$. Clearly $ X_s=R(X)$. As the
Galois group of  $ \widetilde {\mathbb F}$   over $ {\mathbb   F}$
permutes the $ \lambda _i$ we may even choose $ R\in  {\mathbb
F}[\xi ]$. This implies the required continuity.

There is only one semi-simple orbit $ \gamma  _p$  in $ {\mathfrak
g}_p$  and it is closed. We use Frobenius descent (Theorem
\ref{Frob})  for the map $ (X,v,v^*)\mapsto X_s$  from $
{\mathfrak g}_p \times V \times V^*$  to $ \gamma  _p$.

Fix $ a\in  \gamma  _p$ ;  its fiber is the product of $ V\oplus
V^*$  by the set of nilpotent elements which commute with $ a$. It
is a closed subset of the centralizer  $ {\mathfrak m}={\mathfrak
Z}_{{\mathfrak  g}}(a)$  of $ a$  in $ {\mathfrak g}$. Let $
M=Z_G(a)$  and $ \widetilde  M=Z_{\widetilde  G}(a)$.

Following \cite{SS} let us describe these centralizers.
 Let $P$ be the minimal polynomial of $ a$ ; all its roots are simple. Let $P=P_1\dots P_r$ be
the decomposition of $P$ into (distinct) irreducible factors, over
${\mathbb    F}$. If $V_i=\Ker    P_i(a)$, then $V=\oplus V_i$ and
$V^*=\oplus   V_i^*$. An element $x$ of $G$ which commutes with
$a$ is given by  a family $\{x_1,\dots ,x_r\}$ where each $x_i$ is
a linear map from $V_i$ to $V_i$, commuting with the restriction
of $a$ to $V_i$. Now ${\mathbb    F}[\xi ]$ acts on $V_i$, by
specializing $\xi $ to $a_{|V_i}$ and $P_i$ acts trivially so
that, if ${\mathbb    F}_i={\mathbb    F}[\xi ]/(P_i)$, then $V_i$
becomes a vector space over ${\mathbb F}_i$. The ${\mathbb
F}$-linear map $x_i$ commutes with $a$ if and only if it is
${\mathbb F}_i$-linear.\hfill\break

Fix $i$. Let $\ell$ be a non-zero ${\mathbb    F}$-linear form on
${\mathbb    F}_i$. If $v_i\in V_i$ and $v'_i\in V_i^*$ then
$\lambda \mapsto \langle  \lambda v_i,v'_i\rangle  $ is an
${\mathbb    F}$-linear form on ${\mathbb    F}_i$, hence there
exists a unique element $S(v_i,v'_i)$ of ${\mathbb    F}_i$ such
that $\langle  \lambda v_i,v'_i\rangle  =\ell\left  (\lambda
S(v_i,v'_i)\right  )$. One checks trivially that $S$ is ${\mathbb
F}_i$-linear with respect to each variable and defines a non
degenerate duality, over ${\mathbb    F}_i$ between $V_i$ and
$V_i^*$. Here ${\mathbb    F}_i$ acts on $V_i^*$ by transposition,
relative to the ${\mathbb    F}$-duality $\langle  .,.\rangle $,of
the action on $V_i$. Finally if $x_i\in  \End _{{\mathbb F}_i}
V_i$, its transpose, relative to the duality $S(.,.)$ is the same
as its transpose relative to the duality $\langle .,.\rangle $.

Thus $M$ is a product of linear groups and the situation $(M,V,V^*)$ is a composite case, each component being
a linear case (over various extensions of ${\mathbb    F}$).

Let $ u$  be an isomorphism of $ V$  onto $ V^*$  such that $
^ta=uau^{-1}$  and that, for each $ i,\,\,\, u(V_i)=V_i^*$. Then $
u\in  \widetilde  M$  and $ \widetilde  M=M\cup uM$.

Suppose that $ a$  does not belong to the center of $ {\mathfrak
g}$. Then each $ V_i$  has dimension strictly smaller than $ n$
and we can use the inductive assumption. Therefore $ {\mathcal
S}'({\mathfrak  m}\oplus V\oplus V^*)^{\widetilde  M, \chi
}=(0)$. However the nilpotent cone $ {\mathcal  N}_{\mathfrak  m}$
in $ {\mathfrak  m}$  is a closed subset so $ {\mathcal
S}'({\mathcal N}_{\mathfrak  m} \times V \times V^*)^{\widetilde
M, \chi }=(0)$.

Together with Lemma \ref{SuppTinGamma} this proves the first
assertion of the Proposition.

If $ a$  belongs to the center then $ \widetilde  M=\widetilde  G$
and the fiber is $ (a+{\mathcal  N}) \times V \times V^*$. This
implies the second assertion. \end{proof}


\begin{remark} Strictly speaking the singular set is
defined as the set of all $(X,v,v^*)$ such that for  any
polynomial $P$ invariant under $\widetilde G$ one has
$P(X,v,v^*)=P(0)$. Thus, in principle, we also need to consider
the polynomials $P(X,v,v^*)=\langle v^*,X^pv\rangle$ for $p>0$. In
fact, one can show that the support of the distribution $T$ is
contained in the singular set in the strict sense (i.e., the above
polynomials vanish on the support). As this is not needed in the
sequel we omit the proof.
\end{remark}


\section{End of the proof for GL(n)}
\label{SecEndGL} In this section we consider a distribution $ T\in
{\mathcal S}'({\mathcal  N} \times \Gamma )^{\widetilde  G, \chi
}$  and prove that $ T=0$. The following observation will play a
crucial role.

Choose a non-trivial additive character $\psi$ of $\mathbb F$. On $V\oplus V^*$ we have the bilinear form
$$\bigl ((v_1,v_1^*),(v_2,v_2^*)\bigr )\mapsto \langle v_1^*,v_2\rangle +\langle v_2^*,v_1\rangle $$
Define the Fourier transform of a function $\varphi$ on $V\oplus V^*$ by
$$\widehat \varphi (v_2,v_2^*)=\int _{V\oplus V^*}\varphi (v_1,v_1^*)\, \psi (\langle v_1^*,v_2\rangle +\langle v_2^*,v_1\rangle )\, dv_1dv_1^*$$
where $dv_1dv_1^*$ is the self-dual Haar measure.

This Fourier transform commutes with the action of $\widetilde G$;
hence the (partial) Fourier transform $\widehat T$ of our
distribution $T$ has the same invariance properties and the same
support conditions as $T$ itself.







Let $ {\mathcal  N}_i$  be the union of nilpotent orbits of
dimension at most $ i$. We will prove, by descending  induction on
$ i$, that the support of any $(\widetilde{G},\chi)$-equivariant
distribution on $ [{\mathfrak  g}, {\mathfrak  g}] \times \Gamma $
must be contained in $ {\mathcal  N}_i \times \Gamma $. Suppose we
already know that, for some $ i$, the support must be contained in
$ {\mathcal N}_i \times \Gamma $. We must show that, for any
nilpotent orbit $ {\mathcal  O}$  of dimension $ i$, the
restriction of the distribution to $ {\mathcal  O} \times \Gamma $
is 0.

If $ v\in  V$  and $ v^*\in  V^*$  we call $ X_{v,v^*}$  the rank
one map $ x\mapsto \langle v^*,x\rangle v$. Let
$$ \nu_\lambda (X,v,v^*)=(X+\lambda X_{v,v^*},v,v^*),\quad (X,v,v^*)\in  {\mathfrak  g} \times \Gamma ,\,\,\, \lambda \in  {\mathbb   F}.$$
Then $ \nu_\lambda $  is a one parameter group of homeomorphisms
of $ {\mathfrak  g} \times \Gamma $  and note that $ [{\mathfrak
g},{\mathfrak  g}] \times \Gamma $  is invariant. The key
observation is that  {\it $ \nu_\lambda $  commutes with the
action of $ \widetilde  G$ }. Therefore the image of $ T$  by $
\nu_\lambda $  transforms according to the character $\chi $  of $
\widetilde  G$. Its support is contained in $ [{\mathfrak
g},{\mathfrak  g}] \times \Gamma $ and hence must be contained in $
{\mathcal  N} \times \Gamma $  and in fact in $ {\mathcal  N}_i
\times \Gamma $. This means that if $ (X,v,v^*)$  belongs to the
support of $ T$  then, for all $ \lambda $, $ (X+\lambda
X_{v,v^*},v,v^*)$  must belong to $ {\mathcal  N}_i \times \Gamma
$.

The orbit $ {\mathcal  O}$  is open in $ {\mathcal  N}_i$.  Thus
if $ X\in  {\mathcal  O}$  the condition $ X+\lambda X_{v,v^*}\in
{\mathcal  N}_i$  implies that, at least for $ |\lambda |$  small
enough, $ X+\lambda X_{v,v^*}\in  {\mathcal  O}$. It follows that
$ X_{v,v^*}$  belongs to the tangent space to $ {\mathcal  O}$  at
the point $ X$ ; this tangent space is the image of $ \ad   X$.

Define $Q(X)$ to be the set of all pairs $(v,v^*)$ such
$X_{v,v^*}\in \Im \, \ad X$.

By the discussion above, it is enough to prove the following Lemma:
\begin{lemma} \label{KeyLem}
 Let $ T\in  {\mathcal  S}'({\mathcal  O} \times V \times V^*)^{\widetilde  G, \chi  }$.
 Suppose that the support of $ T$  and of $ \widehat T$  are contained    in the set of triplets  $ (X,v,v^*)$  such that $ (v,v^*)\in  Q(X)$.
Then $ T=0$.
\end{lemma}
Note that the trace of $ X_{v,v^*}$  is $ \langle v^*,v\rangle $
and that $ X_{v,v^*}\in  \Im \ad \,  X$ implies that its trace is
0. Therefore $ Q(X)$  is contained in $ \Gamma $.

We proceed in three steps. First we transfer the problem to $
V\oplus V^*$  and a fixed nilpotent endomorphism $ X$. Then we
show that if  Lemma \ref{KeyLem} holds for $ (V_1,X_1)$  and $
(V_2,X_2)$ then it holds for the direct sum $ (V_1\oplus
V_2,X_1\oplus X_2)$. Finally, decomposing $ X$ into Jordan blocks
we are left with the case of a principal nilpotent element for
which we give a direct proof, using Weil representation.

Consider the map $ (X,v,v^*)\mapsto X$  from $ {\mathcal  O}
\times V \times V^*$  onto $ {\mathcal  O}$. Choose $X\in{\mathcal
O}$ and let $ C$ (resp $ \widetilde  C$)  be the stabilizer in $
G$ (resp. in $ \widetilde  G$)  of an element $ X$  of $ {\mathcal
O}$ ; both groups are unimodular, hence we may use Frobenius
descent (Theorem \ref{Frob}).

 Now we have to deal with a distribution, which we still call $T$, which belongs to $   {\mathcal  S}'(V\oplus V^*)^{\widetilde  C, \chi  }$
 such that both $ T$  and its Fourier transform are supported by
 $Q(X)$
 (Proposition \ref{FrFouCommute}). Let us say that $ X$  is \emph{nice} if the only such  distribution is 0.
 We want to prove that all nilpotent endomorphisms are nice.

 \begin{lemma} \label{ProdNiceNice}
  Suppose that we have a decomposition $ V=V_1\oplus V_2$  such that $ X(V_i)\subset V_i$. Let $ X_i$  be the restriction of $ X$  to $ V_i$. Then if $ X_1$  and $ X_2$  are nice, so is $ X$.
 \end{lemma}
\begin{proof}  Let $ (v,v^*)\in  Q(X)$  and choose $ A\in  {\mathfrak  g}$
such that $ X_{v,v^*}=[A,X]$. Decompose $ v=v_1+v_2,\,\,
v^*=v_1^*+v_2^*$  and put
 $$ A=\begin{pmatrix}A_{1,1}&A_{1,2}\cr A_{2,1}&A_{2,2}\end{pmatrix}.$$
 Writing $ X_{v,v^*}$  as a 2 by 2 matrix and looking at the diagonal blocks one gets that $ X_{v_i,v_i^*}=[A_{i,i},X_i]$. This means that
 $$ Q(X)\subset Q(X_1) \times Q(X_2).$$
 For $ i=1,2$  let $ C_i$  be the centralizer of $ X_i$  in $ GL(V_i)$  and $ \widetilde  C_i$  the corresponding extension by $ S_2$.
 Let $ T$  be a distribution as above and let $ \varphi _2\in  {\mathcal  S}(V_2\oplus V_2^*)$. Let $ T_{1}$  be the distribution on $ V_1\oplus V_1^*$  defined by $ \varphi _1\mapsto \langle T,\varphi _1\otimes \varphi _2\rangle $.
 The support of $ T_{1}$  is contained in $ Q(X_1)$  and $ T_{1}$  is invariant under the action of  $ C_1$ . We have
 $$ \langle \widehat T_{1},\varphi _1\rangle =\langle T_{1},\widehat \varphi _1\rangle =\langle T, \widehat {\varphi _1}\otimes \varphi _2\rangle =\langle \widehat T,\check {\varphi _1}\otimes \widehat{\varphi _2}\rangle .$$
 Here $ \check \varphi _1(v_1,v_1^*)=\varphi _1(-v_1,-v_1^*)$.
 By assumption the support of $ \widehat T$  is contained in $ Q(X)$  so that the support of $ \widehat{T_1}$  is supported in $ -Q(X_1)=Q(X_1)$.
 Because $ (X_1)$  is nice this implies that $ T_1$  in invariant under $ \widetilde  C_1$.

 Extend the action of $\widetilde  C_1$ to $V \oplus V^*$
 trivially. We obtain that $T$ is invariant with respect to $\widetilde
 C_1$.  Similarly it is invariant under $ \widetilde  C_2$. Since
 the actions of $\widetilde
 C_1$ and $\widetilde
 C_2$ together with the action of $C$ generate the action of $\widetilde
 C$ we obtain that $ T$
 must be invariant under $ \widetilde  C$ and  hence must be 0.
\end{proof}

 Decomposing $ X$  into Jordan blocks we still have to prove Lemma \ref{KeyLem} for a principal nilpotent element. We need some preliminary results.



 \begin{lemma}
 The distribution $ T$  satisfies the following homogeneity condition:
$$ \langle T,f(tv,tv^*)\rangle =|t|^{-n}\langle T,f(v,v^*)\rangle .$$
\end{lemma}
\begin{proof} We use a particular case of Weil or oscillator
representation. Let $ E$  be a vector space over $ {\mathbb   F}$
of finite dimension $ m$. To simplify assume that $ m$  is even.
Let $ q$ be a non-degenerate quadratic form on $ E$  and let $ b$
be the bilinear form
    $$ b(e,e')=q(e+e')-q(e)-q(e').$$
Fix a continuous non-trivial additive character $ \psi  $  of $
{\mathbb   F}$. We define the Fourier transform on $ E$  by
$$ \widehat f(e')=\int  _Ef(e)\psi  (b(e,e'))de$$
where $ de$  is the self dual Haar measure.

There exists (\cite{RS1}) a representation $ \pi $  of $ SL(2,{\mathbb   F})$  in $ {\mathcal  S}(E)$  such that:
 \begin{eqnarray*}
\pi \begin{pmatrix}1&u\cr 0&1\end{pmatrix}f(e)&=&\psi  (uq(e))f(e)\\
\pi \begin{pmatrix}t&0\cr 0&t^{-1}\end{pmatrix}f(e)&=&\frac{\gamma  (q)}{\gamma  (tq)}|t|^{m/2}f(te)\\
\pi \begin{pmatrix}0&1\cr -1&0\end{pmatrix}f(e)&=&\gamma  (q)\widehat f(e)\\
\end{eqnarray*}
where $\gamma(\cdot)$ is a certain roots of unity, which is $1$ if
$(E,q)$ is a sum of hyperbolic planes.

We have a contragredient action in the dual space $ {\mathcal
S'(E)}$.

Suppose that $ T$  is a distribution on $ E$  such that $ T$  and
$ \widehat T$  are supported on the isotropic cone $ q(e)=0$. This
means that
$$ \langle T,\pi \begin{pmatrix}1&u\cr 0&1\end{pmatrix}f\rangle =\langle T,f\rangle ,\quad \langle \widehat T,\pi \begin{pmatrix}1&u\cr 0&1\end{pmatrix}f\rangle =\langle \widehat T,f\rangle . $$
Using the relation
$$ \langle \widehat T,\varphi \rangle =\langle T,\overline {\gamma  (q)}\pi
\begin{pmatrix}0&1\cr -1&0\end{pmatrix}f\rangle $$
the second relation is equivalent to
$$ \langle T,\pi \begin{pmatrix}1&0\cr -u&1\end{pmatrix}f\rangle =\langle T,f\rangle .$$
The matrices
$$ \begin{pmatrix}1&u\cr 0&1\end{pmatrix},\quad {\text{and}}\quad
\begin{pmatrix}1&0\cr u&1\end{pmatrix}\quad u\in  {\mathbb   F}$$
generate  the group $ SL(2,{\mathbb   F})$. Therefore the
distribution $ T$  is invariant by $ SL(2,{\mathbb   F)}$. In
particular
$$ \langle T,f(te)\rangle =\frac{\gamma  (tq)}{\gamma  (q)}|t|^{-m/2}\langle T,f\rangle $$
and $ T=\gamma  (q)\widehat T$.

\begin{remark} Note that (for even $m$) $\gamma(tq)/\gamma(q)$ is
a character of $t$ and non-zero distributions which are invariant
under $ SL(2,{\mathbb   F})$ do exist. In the case where $m$ is
odd one obtains a representation of the two-fold covering of $
SL(2,{\mathbb   F})$ and we obtain the same homogeneity condition.
However $ \gamma (tq)/\gamma  (q)$  is not a character; hence no
non-zero $T$ can exist.
\end{remark}

In our situation we take $ E=V\oplus V^*$  and $ q(v,v^*)=\langle
v^*,v\rangle $. Then
$$ b\Bigl ((v_1,v_1^*),(v_2,v_2^*)\Bigr )=\langle v_1^*,v_2\rangle +\langle v_2^*,v_1\rangle.$$
The Fourier transform commutes with the action of $ \widetilde
G$. Both $ T$  and $ \widehat T$   are supported on $ Q(X)$  which
is contained in $ \Gamma $. As $ \gamma  (tq)=1$ for all $t$ this
proves the Lemma and also that $ T=\widehat T$.
\end{proof}


\begin{remark} The same type of argument could have been
used for the quadratic form $ \tr(XY)$  on $ {\mathfrak
s}{\mathfrak  l}(V)=[{\mathfrak  g},{\mathfrak  g}]$. This would
have given a short proof for even $ n$  and a homogeneity
condition for odd $ n$.
\end{remark}

Now we find $ Q(X)$.
\begin{lemma}
 If $ X$  is principal then $ Q(X)$  is the set of pairs $ (v,v^*)$  such that for $ 0\leq k<n$, $ \langle v^*,X^kv\rangle =0$.
\end{lemma}
\begin{proof} Choose a basis $ (e_1,\dots ,e_n)$  of $ V$  such that $
Xe_1=0$ and $ Xe_j=e_{j-1}$  for $ j\geq 2$. Consider the map $
A\mapsto XA-AX$  from the space of n by n matrices into itself.
This map is anti-symmetric with respect to the Killing form and
hence its image is the orthogonal complement to its kernel. A
simple computation shows that the kernel of this map, that is to
say the Lie algebra $ {\mathfrak  c}$  of the centralizer  $ C$,
is the space of polynomials (of degree at most $ n-1$ ) in $ X$.
Therefore
%
$$Q(X)=\{(v,v^*) | X_{v,v^*} \in \Im \, \ad X\} =
\{(v,v^*) | \forall \, 0\leq k<n, \, \tr(X_{v,v^*} X^k)=0\}=$$
%
$$=\{(v,v^*) | \forall \, 0\leq k<n, \, \langle
v^*,X^kv\rangle =0 \}.$$
%
\end{proof}

{\it End of the proof of Lemma \ref{KeyLem} }For a principal $X$,
we proceed by induction on $ n$. Keep the above notation. The
centralizer $ C$ of $ X$  is the space of polynomials (of degree
at most $ n-1$) in $ X$  with non-zero constant term. In
particular the orbit $ \Omega $  of $ e_n$  is the open subset $
x_n\ne 0$. We shall prove that the restriction of $ T$  to $
\Omega  \times V^*$ is 0. Note that the centralizer of $ e_n$  in
$ C$  is trivial. By Frobenius descent (Theorem \ref{Frob}), to
the restriction of $ T$ corresponds a distribution $ R$  on $ V^*$
with support in the set of $ v^*$ such that $ (e_n,v^*)\in  Q(X)$.
By the last Lemma this means that $ R$  is a multiple $ a\delta $
of the Dirac measure at the origin. The distribution $ T$
satisfies the two conditions
$$ \langle T,f(v,v^*)\rangle =\langle T,f(tv,t^{-1}v^*)\rangle =|t|^n\langle T,f(tv,tv^*)\rangle .$$
therefore
$$ \langle T,f(v,t^2v^*)=|t|^{-n}\langle T,f(v,v^*)\rangle .$$


 Now $ T$   is recovered from $ R$  by the formula
$$ \langle T,f(v,v^*)=\int  _C\langle R,f(ce_n,^tc^{-1}v^*\rangle  dc=a\int  _Cf(ce_n,0)dc,\,\,\, f\in  {\mathcal  S}(\Omega  \times V^*).$$
Unless $ a=0$  this is not compatible with this last homogeneity condition.

Exactly in the same way one proves that $ T$  is 0 on  $ V \times
\Omega ^*$  where $ \Omega ^*$  is the open orbit $ x_1^*\ne 0$ of
$ C$  in $ V^*$. The same argument is valid for $ \widehat T$
(which is even equal to $ T$ \dots ).

 If $ n=1$  then $ T$  is obviously 0. If $ n\geq 2$  then there exists a distribution $ T'$  on
$$ \bigoplus_{1<j<n}{\mathbb   F}e_j\oplus{\mathbb   F}e_j^*$$
such that,
 $$ T=T'\otimes \delta _{x_n=0}\otimes dx_1\otimes \delta _{x_1^*=0}\otimes dx_n^*.$$
Let $ u$  be the isomorphism of $ V$  onto $ V^*$  given by $
u(e_j)=e_{n+1-j}^*$. Recall that it acts on $ {\mathfrak  g}
\times V \times V^*$  by $ (X,v,v^*)\mapsto
(^t(uXu^{-1}),^tu^{-1}v^*,uv)$. It belongs to $ \widetilde  C$ but
not to $ C$  so it must transform $ T$  into $ -T$.

The case $ n=1$  has just been settled. If $ n=2$   in the above
formula $ T'$  should be replaced by a constant. The constant must
be 0 if we want $ u(T)=-T$. If $ n>2$ let
$$V'=\Bigl (\oplus_1^{n-1}{\mathbb F}e_i\Bigr )/{\mathbb F}e_1$$
and let $X'$ be the nilpotent endomorphism of $V'$ defined by $X$.
We may consider $T'$ as a distribution on $V' \oplus V'^{*}$ and
one easily checks that, with obvious notation, it transforms
according to the character $\chi$ of the the centralizer
$\widetilde C'$ of $X'$ in $\widetilde G'$. By induction $T'=0$,
hence $T=0$.

\hfill$\Box$\hfill

\section{Reduction to the singular set: the orthogonal and unitary
cases} \label{SecRedSingSetO}



We now turn our attention to the unitary case. We keep the
notation of the introduction. In particular $W=V\oplus {\mathbb
D}e$ is a vector space over ${\mathbb  D}$ of dimension $n+1$ with
a non-degenerate hermitian form $\langle.,.\rangle $ such that $e$
is orthogonal to $V$. The unitary group $G$ of $V$ is embedded
into the unitary group  $M$ of $W$.

Let $A$ be the set of all  bijective maps $u$ from $V$ to $V$ such that
$$ u(v_1+v_2)=u(v_1)+u(v_2),\,\,\, u(\lambda v)=\overline\lambda u(v),\,\,\, \langle u(v_1),u(v_2)\rangle =\overline{\langle v_1,v_2\rangle }.$$
An example of such a map is obtained by choosing a basis $e_1,\dots ,e_n$ of $V$ such that $\langle e_i,e_j\rangle \in {\mathbb  F}$ and defining
$$ u(\sum x_ie_i)=\sum \overline x_ie_i.$$
Any $u\in A$ is extended to $W$ by the rule $u(v+\lambda
e)=u(v)+\overline\lambda e$ and we define an action on $\GL (W)$
by $m\mapsto um^{-1}u^{-1}$. The group $G$ acts on $\GL (W)$ by
conjugation.

Let $\widetilde G$ be the group of bijections of $\GL(W)$ onto
itself generated by the actions of $G$  and $A$. It is a semi
direct product of $G$ and $S_2$. We identify $G$ with a subgroup
of $\widetilde G$ and $A$ with $\widetilde G\setminus G$. Note
that $\widetilde G$ preserves $M$. When a confusion is possible we
denote the product in $\widetilde G$ by $\times$.




  We define a character $\chi $ of $\widetilde G$ by $\chi (g)=1$ for $g\in G$ and $\chi (u)=-1$ for $u\in \widetilde G\setminus G$.
 Our overall goal is to prove that ${\mathcal  S}'(M)^{\widetilde G,\chi }=(0)$.

 Let $\widetilde G$ act on $G\times V$ as follows:
 $$ g(x,v)=(gxg^{-1},g(v)),\,\, u(x,v)=(ux^{-1}u^{-1},-u(v)),\quad g\in G,u\in A,x\in G,v\in V$$
Our first step is to replace $M$ by $G\times V$.
\begin{proposition}
 Suppose that for any $V$ and any hermitian form ${\mathcal  S}'(G\times V)^{\widetilde G,\chi }=(0)$, then ${\mathcal  S}'(M)^{\widetilde G,\chi }=(0)$.
\end{proposition}
\begin{proof} We have in particular ${\mathcal  S}'(M\times
W)^{\widetilde M,\chi }=(0)$. Let $Y$ be the set of all $(m,w)$
such that $\langle w,w\rangle =\langle e,e\rangle $; it is a
closed subset, invariant under $\widetilde M$, hence ${\mathcal
S}'(Y)^{\widetilde M,\chi }=(0)$. By Witt's theorem $M$ acts
transitively on $\Gamma =\{w|\langle w,w\rangle =\langle
e,e\rangle \} $. We can apply Frobenius descent (Theorem
\ref{Frob}) to the map $(m,w)\mapsto w$ of $Y$ onto $\Gamma $. The
centralizer of $e$ in $\widetilde M$ is isomorphic to $\widetilde
G$ acting as before on the fiber  $M\times \{e\}$. We have a
linear bijection between ${\mathcal  S}'(M)^{\widetilde G.\chi }$
and ${\mathcal S}'(Y)^{\widetilde M,\chi }$; therefore ${\mathcal
S}'(M)^{\widetilde G.\chi }=(0)$. \end{proof}

The proof that ${\mathcal  S}'(G\times V)^{\widetilde G,\chi }=(0)$ is by induction on $n$. If ${\mathfrak  g}$ is the Lie algebra of $G$ we shall prove simultaneously that ${\mathcal  S}'({\mathfrak  g}\times V)^{\widetilde G,\chi }=(0)$. In this case $G$ acts on its Lie algebra by the adjoint action and for $u\in \widetilde G\setminus G$ one puts, for $X\in {\mathfrak  g},\,\,$ $u(X)=-uXu^{-1}$.


The case $n=0$ is trivial so we may assume that $n\geq 1$. If $T\in {\mathcal  S}'(G\times V)^{\widetilde G,\chi }$ in this section we will prove that the support of $T$ must be contained in the "singular set".

Let $Z$ (resp. {$\mathfrak  z$}) be the center of $G$ (resp. {$\mathfrak  g$}) and ${\mathcal  U}$ (resp. {$\mathcal  N$}) the (closed) set of all unipotent (resp. nilpotent) elements of $G$ (resp. {$\mathfrak  g$}).
\begin{lemma} \label{HCO}
 If $T\in {\mathcal  S}'(G\times V)^{\widetilde G,\chi }$ (resp. $T\in {\mathcal  S}'({\mathfrak  g}\times V)^{\widetilde G,\chi }$) then the support of $T$ is contained in $Z{\mathcal  U}\times V$ (resp. $({\mathfrak  z}+{\mathcal  N})\times V)$.
\end{lemma}
\begin{proof}
This is Harish-Chandra's descent. We first review some facts about
the centralizers of semi-simple elements, following \cite{SS}.

 Let $a\in G$, semi-simple; we want to describe its centralizer $G_a$ (resp. $\widetilde G_a$) in $G$ (resp. in $\widetilde G$) and to show that ${\mathcal  S}'(G_a\times
V)^{\widetilde G_a,\chi }=(0)$.

 View $a$ as a ${\mathbb  D}$-linear endomorphism of $V$ and call $P$ its minimal polynomial. Then, as $a$ is semi-simple,
 $P$ decomposes into distinct irreducible factors $P=P_1\dots P_r$. Let $V_i=\Ker P_i(a)$ so that $V=\oplus V_i$.
 Any element $x$ which commutes with $a$ will satisfy $xV_i\subset V_i$ for each $i$.
For
$$ R(\xi )=d_0+\cdots +d_m\xi^m,\quad d_0d_m\ne  0$$
let
$$ R^*(\xi )=\overline{d_0}\xi ^m+\cdots +\overline{d_m}.$$
Then, from $aa^*=1$ we obtain, if $m$ is the degree of $P$
$$ \langle P(a)v,v'\rangle =\langle v,a^{-m}P^*(a)v'\rangle $$
(note that the constant term of $P$ can not be 0 because $a$ is
invertible). It follows that $P^*(a)=0$ so that $P^*$ is
proportional to $P$. Now $P^*=P_1^*\dots P_r^*$; hence there exists
a bijection $\tau $ from $\{1,2,\dots ,r\}$ onto itself such that
$P^*_i$ is proportional to $P_{\tau (i)}$. Let $m_i$ be the degree
of $P_i$. Then, for some non-zero constant $c$
$$ 0=\langle P_i(a)v_i,v_j\rangle =\langle v_i,a^{-m_i}P_i^*(a)v_j\rangle =c\langle v_i,a^{-m_i}P_{\tau (i)}(a)v_j\rangle
,\quad v_i\in V_i,\,\, v_j\in V_j.$$
We have two possibilities.

{\bf Case 1:}$\,\, \tau (i)=i$. The space $V_i$ is orthogonal to
$V_j$ for $j\ne  i$; the restriction of the hermitian form to
$V_i$ is non-degenerate. Let ${\mathbb  D}_i={\mathbb  D}[\xi
]/(P_i)$ and consider $V_i$ as a vector space over ${\mathbb D}_i$
through the action $(R(\xi ),v)\mapsto R(a)v$. As $a_{|V_i}$ is
invertible, $\xi $ is invertible modulo $(P_i)$; choose $\eta$
such that $\xi \eta=1$ modulo $(P_i)$. Let $\sigma _i$ be the
semi-linear involution of ${\mathbb  D}_i$, as an algebra over
${\mathbb  D}$:
$$ \sum d_j\xi^j\mapsto \sum \overline{d_j}\eta^j\pmod P_i$$

Let ${\mathbb  F}_i$ be the subfield of fixed points for
$\sigma_i$. It is a finite extension of ${\mathbb  F}$, and
${\mathbb  D}_i$ is either a quadratic extension of ${\mathbb
F}_i$ or equal to ${\mathbb  F}_i$. There exists a ${\mathbb
D}$-linear form $\ell\ne  0$  on ${\mathbb  D}_i$ such that $\ell
(\sigma_i(d))=\overline{\ell (d)}$ for all $d\in {\mathbb  D }_i$.
Then any ${\mathbb  D}$-linear form $L$ on ${\mathbb  D}_i$ may be
written as $d\mapsto \ell (\lambda d)$ for some unique $\lambda
\in {\mathbb  D}_i$.

If $v,v'\in V_i$ then $d\mapsto \langle d(a)v,v'\rangle $ is ${\mathbb  D}$-linear map on ${\mathbb  D}_i$; hence there exists $S(v,v')\in {\mathbb  D}_i$ such that
$$ \langle d(a)v,v'\rangle =\ell (dS(v,v')).$$
One checks that $S$ is a non-degenerate hermitian form on $V_i$ as
a vector space over ${\mathbb  D}_i$. Also a ${\mathbb  D}$-linear
map $x_i$ from $V_i$ into itself commutes with $a_i$ if and only
if it is ${\mathbb  D}_i$-linear and it is unitary with respect to
our original hermitian form if and only if it is unitary with
respect to $S$. So in this case we call $G_i$  the unitary group
of $S$. It does not depend upon the choice of $\ell$. As no
confusion may arise, for $\lambda \in {{\mathbb  D}_i}$ we define
$\overline\lambda =\sigma_i(\lambda )$.

We choose an ${\mathbb  F}_i$-linear map $u_i$ from $V_i$ onto
itself, such that $u_i(\lambda v)=\overline\lambda u(v)$ and
$S(u_i(v),u_i(v'))=\overline{S(v,v')}$. Then because of our
original choice of $\ell$ we also have $\langle
u_i(v),u_i(v')\rangle =\overline{\langle v,v'\rangle }$. Note that
$u(a_{|V_i})^{-1}u^{-1}=a_{|V_i}$.

{\bf Case 2.} Suppose now that $j=\tau (i)\ne  i$. Then $V_i\oplus V_j$ is orthogonal to $V_k$ for $k\ne  i,j$ and the restriction of the hermitian form to $V_i\oplus V_j$ is non-degenerate, both $V_i$ and $V_j$ being totally isotropic subspaces. Choose an inverse $\eta$ of $\xi $ modulo $P_j$. Then for any $P\in {\mathbb  D}[\xi ]$
$$ \langle P(a)v_i,v_j\rangle =\langle v_i,\overline P(\eta (a))v_j\rangle ,\quad v_i\in V_i,\,\, v_j\in V_j$$
where $\overline P$ is the polynomial obtained from $P$ by conjugating
its coefficients. This defines a map, which
we call $\sigma_i$ from ${\mathbb  D}_i$ onto ${\mathbb  D}_j$. In
a similar way we have a map $\sigma _j$ which is the inverse of
$\sigma_i$. Then, for $\lambda \in {\mathbb  D}_i$ we have
$\langle \lambda v_i,v_j\rangle =\langle v_i,\sigma_i(\lambda
)v_j\rangle $.


 View $V_i$ as a vector space over ${\mathbb  D}_i$. The action
$$ (\lambda ,v_j)\mapsto
 \sigma_i(\lambda )v_j$$
defines a structure of ${\mathbb  D}_i$ vector space on $V_j$.
However note that for $\lambda \in {\mathbb  D}$ we have
$\sigma_i(\lambda )=\overline\lambda $ so that $\sigma_i(\lambda
)v_j$ may be different from $\lambda v_j$. To avoid confusion we
shall write, for $\lambda \in {\mathbb  D}_i$
$$\lambda v_i=\lambda *v_i\quad \text{and}\quad \sigma_i(\lambda )v_j=\lambda *v_j.$$

As in the first case choose a non-zero ${\mathbb  D}$-linear form $\ell$ on ${\mathbb  D}_i$. For $v_i\in V_i$ and $v_j\in V_j$ the map $\lambda \mapsto \langle \lambda *v_i,v_j\rangle $ is a ${\mathbb  D}$-linear form on ${\mathbb  D}_i$; hence there exists a unique element $S(v_i,v_j)\in {\mathbb  D}_i$ such that, for all $\lambda $
$$ \langle \lambda *v_i,v_j\rangle =\ell(\lambda S(v_i,v_j)).$$
The form $S$ is ${\mathbb  D}_i$- bilinear and non-degenerate so that we can view $V_j$ as the dual space over ${\mathbb  D}_i$ of the ${\mathbb  D}_i$ vector space $V_i$.

Let $(x_i,x_j)\in {\End}_{\mathbb  D}(V_i)\times {\End}_{\mathbb
D}(V_j)$. They commute with $(a_i,a_j)$ if and only if they are
${\mathbb  D}_i$-linear. The original hermitian form will be
preserved, if and only if $S(x_iv_i,x_jv_j)=S(v_i,v_j)$ for all
$v_i,v_j$. This means that $x_j$ is the inverse of the transpose
of $x_i$. In this situation we define $G_i$ as the linear group of
the ${\mathbb  D}_i$-vector space $V_i$.

Let $u_i$ be a ${\mathbb  D}_i$-linear bijection of $V_i$ onto $V_j$. Then $u_i(av_i)=a^{-1}u_i(v_i)$ and $u_i^{-1}(av_j)=a^{-1}u_i^{-1}(v_j)$.



Recall that $G_a$ is the centralizer of $a$ in $G$. Then $(G_a,V)$
decomposes as a "product", each "factor" being either of type
$(G_i,V_i)$ with $G_i$ a unitary group (case 1) or $(G_i,V_i\times
V_j)$ with $G_i$ a general linear group (case 2). Gluing together
the  $u_i$ (case 1) and the $(u_i,u_i^{-1})$ (case 2) we get an
element $u\in \widetilde G\setminus G$ such that $ua^{-1}u^{-1}=a$
which means that it belongs to the centralizer of $a$ in
$\widetilde G$. Finally if $\widetilde G_a$ is the centralizer of
$a$ in $\widetilde G$ then $(\widetilde G_a,V)$ is  imbedded into
a product each "factor" being either of type $(\widetilde
G_i,V_i)$ with $G_i$ a unitary group (case 1) or $(\widetilde
G_i,V_i\times V_j)$ with $G_i$ a general linear group (case 2).

If $a$ is not central then for each $i$ the dimension of $V_i$ is
strictly smaller than $n$ and from the result for the general
linear group and the inductive assumption in the  orthogonal or
unitary case we conclude that ${\mathcal  S}'(G_a\times
V)^{\widetilde G_a,\chi }=(0)$.

{\it Proof of Lemma \ref{HCO}} in the group case. Consider the map
$g\mapsto P_g$ where $P_g$ is the characteristic polynomial of
$g$. It is a continuous map from $G$ into the set of polynomials
of degree at most $n$. Each non-empty fiber ${\mathcal  F}$ is
stable under $ G$ but also under $\widetilde G\setminus G$.
Bernstein's localization principle tells us that it is enough to
prove that ${\mathcal  S}'({\mathcal  F}\times V)^{\widetilde
G,\chi }=(0)$.

Now it follows from \cite[chapter IV]{SS}  that ${\mathcal  F}$
contains only a finite number of semi-simple orbits; in particular
the set of semi-simple elements ${\mathcal  F}_s$ in ${\mathcal
F}$ is closed. Let us use the multiplicative Jordan decomposition
into a product of a semi-simple and a unipotent element.  Consider
the map $\theta $ from ${\mathcal  F}\times V$ onto ${\mathcal
F}_s$ which associates to $(g,v)$ the semi-simple part $g_s$ of
$g$. This map is continuous (see the corresponding proof for $GL$)
and commutes with the action of $\widetilde G$. In ${\mathcal
F}_s$ each orbit $\gamma  $ is both open and closed therefore
$\theta ^{-1}(\gamma  )$ is open and closed and invariant under
$\widetilde G$. It is enough to prove that for each such orbit
${\mathcal  S}'(\theta ^{-1}(\gamma  ))^{\widetilde G,\chi }=(0)$.
By Frobenius descent (Theorem \ref{Frob}), if $a\in \gamma  $ and
is not central, this follows from the above considerations on the
centralizer of such an $a$ and the fact that $\theta ^{-1}(a)$ is
a closed subset of the centralizer of $a$ in $\widetilde G$, the
product of the set of unipotent element commuting with $a$ by $V$.
Now $g_s$ is central if and only if $g$ belongs to $Z{\mathcal
U}$, hence the Lemma. For the Lie algebra the proof is similar,
using the additive Jordan decomposition.
\end{proof}

Going back to the group if $a$ is central we see that it suffices to prove that ${\mathcal  S}'({\mathcal  U}\times V)^{\widetilde G,\chi }=(0)$ and similarly for the Lie algebra it is enough to prove that ${\mathcal  S}'({\mathcal  N}\times V)^{\widetilde G,\chi }=(0)$.

Now the exponential map (or the Cayley transform) is a
homeomorphism of ${\mathcal  N}$ onto ${\mathcal  U}$ commuting
with the action of $\widetilde G$. Therefore it is enough to
consider the Lie algebra case.

We now turn our attention to $V$. Let
$$ \Gamma =\{v\in V | \langle v,v\rangle =0\}$$
\begin{proposition}
 If $T\in {\mathcal  S}'({\mathcal  N}\times V)^{\widetilde G,\chi }$ then the support of $T$ is contained in ${\mathcal  N}\times \Gamma $.
\end{proposition}
\begin{proof} Let
$$ \Gamma _t=\{v\in V\,|\, \langle v,v\rangle =t\}$$
Each $\Gamma _t$ is stable by $\widetilde G$, hence, by
Bernstein's localization principle, to prove that the support of
$T$ is contained in ${\mathcal  N}\times \Gamma _0$ it is enough
to prove that, for $t\ne  0$, $\,\, {\mathcal  S}'({\mathcal
N}\times \Gamma _t)^{\widetilde G,\chi }=(0)$.

By Witt's theorem the group $G$ acts transitively on $\Gamma _t$.
We can apply Frobenius descent to the projection from ${\mathcal
N}\times \Gamma _t$ onto $\Gamma _t$. Fix a point $v_0\in \Gamma
_t$. The fiber is ${\mathcal  N}\times \{v_0\}$. Let $\widetilde
G_1$ be the centralizer of $v_0$ in $\widetilde G$. We have to
show that ${\mathcal  S}'({\mathcal  N})^{\widetilde G_1,\chi
}=(0)$ and it is enough to prove that  ${\mathcal  S}'({\mathfrak
g})^{\widetilde G_1,\chi }=(0)$.

The vector $v_0$ is not isotropic so we have an orthogonal decomposition
$$ V={\mathbb  D}v_0\oplus V_1$$
with $V_1$ orthogonal to $v_0$. The restriction of the hermitian form to $V_1$ is non-degenerate and $G_1$ is identified with the unitary group of this restriction, and   $\widetilde G_1$ is the expected semi-direct product with $S_2$. As a $\widetilde G_1$-module the Lie algebra ${\mathfrak  g}$ is isomorphic to a direct sum
$$ {\mathfrak  g\approx {\mathfrak  g}_1\oplus V_1\oplus W}$$
where ${\mathfrak  g}_1$ is the Lie algebra of $G_1$  and $W$ a vector space over ${\mathbb  F}$ of dimension 0 or 1 and on which the action of $\widetilde G_1$ is trivial. The action on ${\mathfrak  g}_1\oplus V_1$ is the usual one so that, by induction, we know that ${\mathcal  S}'({\mathfrak  g}_1\oplus V_1)^{\widetilde G_1,\chi }=(0)$. This readily implies that ${\mathcal  S}'({\mathfrak  g})^{\widetilde G_1,\chi }=(0)$.
\end{proof}

Summarizing: it remains to prove that ${\mathcal S}'({\mathcal N}\times\Gamma )^{\widetilde G,\chi}=(0)$.


\section{End of the proof in the orthogonal and unitary
cases} \label{SecEndO}

We keep our general notation. We have to show that a distribution on ${\mathcal  N}\times \Gamma $ which is invariant under $G$ is invariant under $\widetilde G$. To some extent the proof will be similar to the one we gave for the general linear group.

 In particular we will use the fact that if $T$ is such a distribution then its partial Fourier transform on $V$ is also invariant under $G$. The Fourier transform on $V$ is defined using the bilinear form
 $$ (v_1,v_2)\mapsto \langle v_1,v_2\rangle +\langle v_2,v_1\rangle $$
 which is invariant under $\widetilde G$.

 For $v\in V$ put
 $$ \varphi   _v(x)=\langle x,v\rangle v,\quad x\in V.$$
 It is a rank one endomorphism of $V$ and $\langle \varphi   _v(x),y\rangle =\langle x,\varphi   _v(y)\rangle $.
 \begin{lemma}$ $
 \begin{enumerate}
\item In the unitary case, for $\lambda \in {\mathbb  D}$ such that $\lambda =-\overline\lambda $ the map
 $$ \nu_\lambda : \quad (X,v)\mapsto (X+\lambda \varphi   _v,v)$$
 is a homeomorphism of $[{\mathfrak  g},{\mathfrak  g}]\times \Gamma $ onto itself which commutes with $\widetilde G$.\hfill\break
\item In the orthogonal case, for $\lambda \in {\mathbb  F}$ the
map
 $$ \mu  _\lambda :\quad (X,v)\mapsto (X+\lambda X\varphi   _v+\lambda \varphi   _vX,v)$$
  is a homeomorphism of $[{\mathfrak  g},{\mathfrak  g}]\times \Gamma $ onto itself which commutes with $\widetilde G$.
\end{enumerate}
 \end{lemma}
 The proof is a trivial verification.

 We now use the stratification of ${\mathcal  N}$. Let us first check that
 a $G$-orbit is stable by $\widetilde G$.
\footnote{In fact, we only need this for nilpotent orbits and this
will be done later in an explicit way, using the canonical form of
nilpotent matrices.}

 Choose a basis $e_1,\dots ,e_n$ of $V$ such that $\langle e_i,e_j\rangle \in {\mathbb  F}$;
 this gives a conjugation $u: v=\sum x_ie_i\mapsto \overline v=\sum \overline{x_i}e_i$ on $V$.
 If $A$ is any endomorphism of $V$ then $\overline A$ is the endomorphism $v\mapsto \overline{A(\overline v)}$.
 The conjugation $u$ is an element of $\widetilde G\setminus G$ and, as such, it acts on ${\mathfrak  g}\times V$ by
 $(X,v)\mapsto (-uXu^{-1},-u(v))=(-\overline X,-\overline v)$. In \cite[Chapter 4, Proposition
 1-2]{MVW}
it is shown that for $X\in {\mathfrak  g}$ there exists an
${\mathbb  F}$-linear automorphism $a$ of $V$ such that $\langle
a(x),a(y)\rangle =\overline{\langle x,y\rangle }$ (this implies
that $a(\lambda x)=\overline{\lambda }x$) and such that
$aXa^{-1}=-X$. Then $g=ua\in G$ and $gXg^{-1}=-\overline X$ so
that $-\overline X$ belongs to the $G$-orbit of $X$. Note that
$a\in \widetilde G\setminus G$ and as such acts as
$a(X,v)=(X,-a(v))$; it is an element of the centralizer of $X$ in
$\widetilde G\setminus G$.

Let ${\mathcal  N}_i$ be the union of all nilpotent orbits of
dimension at most $i$. We shall prove, by descending induction on
$i$, that the support of a distribution $T\in {\mathcal
S}'({\mathcal  N}\times \Gamma )^{\widetilde G,\chi }$ must be
contained in ${\mathcal  N}_i\times \Gamma $.



  So now assume that $i\geq 0$ and that we already know that the support of any $T\in {\mathcal  S}'({\mathcal  N}\times \Gamma )^{\widetilde G,\chi }$ must be contained in ${\mathcal  N}_i\times \Gamma $. Let ${\mathcal  O}$ be a nilpotent orbit of dimension $i$; we have to show that the restriction of $T$ to ${\mathcal  O}$ is 0.

In the unitary case fix $\lambda \in {\mathbb  D}$ such that
$\lambda =-\overline\lambda $ and consider,  for every $t\in
{\mathbb F}$ the homeomorphism $\nu_{t\lambda }$; the image of $T$
belongs to ${\mathcal  S}'({\mathcal  N}\times \Gamma
)^{\widetilde G,\chi }$ so that the image of the support of $T$
must  be contained in ${\mathcal  N}_i\times \Gamma $. If $(X,v)$
belongs to this support this means that $X+t\lambda \varphi _v\in
{\mathcal N}_i$.

If $i=0$ so that $\mathcal {N}_i=\{0\}$ this implies that  $v=0$ so that $T$ must be a multiple of the Dirac measure at the point $(0,0)$ and hence is invariant under $\widetilde G$ so must be 0.

If $i>0$ and $X\in {\mathcal  O}$ then as ${\mathcal  O}$ is open
in ${\mathcal  N}_i$, we get that, at least for $|t|$ small
enough, $X+t\lambda \varphi   _v\in {\mathcal  O}$ and therefore
$\lambda \varphi   _v$ belongs to the tangent space $\Im \,\ad
(X)$ of ${\mathcal  O}$ at the point $X$. Define
$$ Q(X)=\{v\in V|\varphi   _v\in \Im \, \ad (X)\},\quad X\in {\mathcal  N},\,\,\, \text {(unitary case)}.$$
Then we know that the support of the restriction of $T$ to ${\mathcal  O}$ is contained in
$$ \{(X,v)|X\in {\mathcal  O,\, v\in Q(X)}\}$$
and the same is true for the partial Fourier transform of $T$ on $V$.

In the orthogonal case for $i=0$, the distribution $T$ is the
product of the Dirac measure at the origin of $\mathfrak{g}$ by  a
distribution $T'$ on $V$. The distribution $T'$ is invariant under
$G$ but the image of $\widetilde G$ in $\End (V)$ is the same as
the image of $G$ so that $T'$ is invariant under $\widetilde G$
hence must be 0.

If $i>0$ we proceed  as in the unitary case, using $\mu  _\lambda $. We define
$$ Q(X)=\{v\in V|X\varphi   _v+\varphi   _vX\in \Im \, \ad (X)\},\quad X\in {\mathcal  N},\,\,\, {\text {(orthogonal case)}}$$
and we have the same conclusion.

In both cases, for $i>0$, fix $X\in {\mathcal  O}$. We use
Frobenius descent for the projection  map  $(Y,v)\mapsto Y$ of
${\mathcal  O}\times V$ onto ${\mathcal  O}$. Let $C$ (resp.
$\widetilde C$) be the stabilizer of $X$ in $G$ (resp. $\widetilde
G$). We have a linear bijection of ${\mathcal  S}'({\mathcal
O}\times \Gamma )^{\widetilde G,\chi }$ onto ${\mathcal
S}'(V)^{\widetilde C,\chi }$.
\begin{lemma}
  Let $T\in {\mathcal  S}'(V)^{\widetilde C,\chi }$. If $T$ and its Fourier transform are supported in $Q(X)$ then $T=0$.
\end{lemma}
Let us say that a nilpotent element $X$ is nice if the above Lemma is true.

Suppose that we have a direct sum decomposition $V=V_1\oplus V_2$
such that $V_1$ and $V_2$ are orthogonal. By restriction we get
non-degenerate hermitian forms $\langle .,.\rangle _i$ on $V_i$.
We call $G_i$ the unitary group of $\langle .,.\rangle _i$,
${\mathfrak  g}_i$ its Lie algebra and so on. Suppose that
$X(V_i)\subset V_i$ so that $X_i=X_{|V_i}$ is a nilpotent element
of ${\mathfrak  g}_i$.
\begin{lemma} \label{ProdNiceNiceO}
If $X_1$ and $X_2$ are nice so is $X$.
\end{lemma}
\begin{proof} We claim that $Q(X)\subset Q(X_1)\times Q(X_2)$. Indeed if
$$ A=\begin{pmatrix}A_{1,1}&A_{1,2}\cr A_{2,1}&A_{2,2}\end{pmatrix}\in {\mathfrak  g}$$
then from
$$ \langle A\begin{pmatrix}x_1\cr x_2\end{pmatrix},
\begin{pmatrix}y_1\cr y_2\end{pmatrix}\rangle +\langle \begin{pmatrix}x_1\cr x_2\end{pmatrix},A\begin{pmatrix}y_1\cr y_2\end{pmatrix}\rangle =0$$
we get in particular
$$ \langle A_{i,i}x_i,y_i\rangle +\langle x_i,A_{i,i}y_i\rangle =0$$
so that $A_{i,i}\in {\mathfrak  g}_i$. Note that
$$ [X,A]=\begin{pmatrix}[X_1,A_{1,1}]&*\cr *&[X_2,A_{2,2}]\end{pmatrix}.$$

If $v_i\in V_i$ and $v_j\in V_j$ we define $\varphi   _{v_i,v_j}: V_i\mapsto V_j$ by $\varphi   _{v_i,v_j}(x_i)=\langle x_i,v_i\rangle v_j$. Then, for $v=v_1+v_2$
$$ \varphi   _{v}=\begin{pmatrix}\varphi   _{v_1,v_1}&\varphi   _{v_2,v_1}\cr \varphi   _{v_1,v_2}&\varphi   _{v_1,v_2}\end{pmatrix}.$$
Therefore if, for $A\in {\mathfrak  g}$ we have $\varphi   _v=[X,A]$ then $\varphi   _{v_i,v_i}=[X_i,A_{i,i}]$. This proves the assertion for the unitary case. The orthogonal case is similar.

The  end of the proof  is the same as the end of the proof of
Lemma \ref{ProdNiceNice}.
\end{proof}

Now in both orthogonal and unitary cases nilpotent elements have normal forms which are orthogonal direct sums of "simple" nilpotent matrices.
This is precisely described in \cite{SS} IV 2-19 page 259. By the above Lemma it is enough to prove that each "simple" matrix is nice.

{\bf Unitary case.} There is only one type to consider.  There
exists a basis $e_1,\dots ,e_n$ of $V$ such that $Xe_1=0$ and
$Xe_i=e_{i-1}, \, i\geq 2$. The hermitian form is given by
$$ \langle e_i,e_j\rangle =0\, \, \text{if}\, i+j\ne  n+1,\quad \langle e_i,e_{n+1-i}\rangle =(-1)^{n-i}\alpha$$
with $\alpha\ne  0$. Note that $\overline\alpha =(-1)^{n-1}\alpha
$. Suppose that $v\in Q(X)$; for some $A\in {\mathfrak  g}$ we
have $\lambda \varphi   _v=XA-AX$. For any integer $p\geq 0$
$${\tr  }  (\lambda \varphi   _vX^p)={\tr} (XAX^{p}-AX^{p+1})=0.$$
Now ${\tr} (\varphi   _vX^p)=\langle X^pv,v\rangle $ Let $v=\sum
x_ie_i$. Hence
$$ \langle X^pv,v\rangle =\sum _1^{n-p}x_{i+p}\langle e_i,v\rangle =\sum _1^{n-p}(-1)^{n-i}\alpha x_{i+p}\overline x_{n+1-i}=0.$$
For $p=n-1$ this gives $x_n\overline x_n=0$. For $p=n-2$ we get
nothing new but for $p=n-3$ we obtain $x_{n-1}=0$. Going on, by an
easy induction, we conclude that $x_i=0$ if $i\geq (n+1)/2$.

If $n=2p+1$ is odd put $V_1=\oplus _1^p{\mathbb  D}e_i$,$\,\, V_0={\mathbb  D}e_{p+1}$ and $V_2=\oplus _{p+2}^{2p+1}{\mathbb  D}e_i$.
If $n=2p$ is even put $V_1=\oplus _1^p{\mathbb  D}e_i$, $V_0=(0)$ and $V_2=\oplus _{p+1}^{2p}{\mathbb  D}e_i$.
In both cases we have $V=V_1\oplus V_0\oplus V_2$. We use the notation $v=v_2+v_0+v_1$

The distribution $T$ is supported by $V_1$. Call $\delta  _i$ the
Dirac measure at $0$ on $V_i$. Then we may write $T=U\otimes
\delta _0\otimes \delta  _2$ with $U\in {\mathcal  S'(V_1)}$. The
same thing must be true of the Fourier transform of $T$. Note that
$\widehat U$ is a distribution on $V_2$, that $\widehat\delta _2$
is a Haar measure $dv_1$ on $V_1$ and that, for $n$ odd
$\widehat\delta _0$ is a Haar measure $dv_0$ on $V_0$. So we have
$\widehat T= dv_1\otimes \widehat U$ if $n$ is even and $\widehat
T=dv_1\otimes dv_0\otimes \widehat U$ if $n$ is odd. In the odd
case this forces $T=0$. In the even case, up to a scalar multiple
the only possibility is $T=dv_1\otimes \delta  _2$.

Let $$ a: \sum x_ie_i\mapsto \sum (-1)^i\overline x_ie_i.$$ Then
$a\in \widetilde G\setminus G$. It acts on ${\mathfrak  g}$ by
$Y\mapsto -aYa^{-1}$ and in particular  $-aXa^{-1}=X$ so that
$a\in \widetilde C\setminus C$. The action on $V$ is given by
$v\mapsto -a(v)$. It is an involution. The subspace $V_1$ is
invariant and so $dv_1$ is invariant. This implies that $T$ is
invariant under $\widetilde C$ so it must be 0.

{\bf Orthogonal case.} There are two different types of "simple"
nilpotent matrices.

{\bf The first type} is the same as the unitary case, with
$\alpha=1$ and thus $n$ odd but now our condition is that
$X\varphi _v+\varphi   _vX=[X,A]$
 for some $A\in {\mathfrak  g}$. As before this implies that ${\tr} (\varphi   _vX^q)=0$ but only for $q\geq 1$.
 Put $n=2p+1$; we get $x_j=0$ for $j>p+1$.
Decompose $V$ as before: $V=V_1\oplus V_0\oplus V_2$. Our
distribution $T$ is supported by the subspace $v_2=0$ so we write
it $T=U\otimes \delta  _2$ with $U\in {\mathcal  S}'(V_1\oplus
V_0)$. This is also true  for the distribution $\widehat T$ so we
must have $U=dv_1\otimes R$ with $R$ a distribution on $V_0$.
Finally $T=dv_1\otimes R\otimes \delta_2$. Now $-\Id \in C$ and
$T$ is invariant under $C$ so that $R$ must be an even
distribution. On the other end the endomorphism $a$ of $V$ defined
by $a(e_i)=(-1)^{i-p-1}e_i$ belongs to $C$ and $aXa^{-1}=-X$ and
$u: (X,v)\mapsto (-X,-v)$ belongs to $\widetilde G\setminus G$.
The product $a \times u$ of $a$ and $u$ in $\widetilde G$ belongs
to $\widetilde C \setminus C$. Clearly $T$ is invariant under
$a\times u$ so that $T$ is invariant under $\widetilde C$ so it
must be 0.

{\bf The second type} is as follows. We have $n=2m$, an even integer and a decomposition $V=E\oplus F$ with both $E$ and $F$ of dimension $m$.
We have a basis $e_1,\dots ,e_m$ of $E$ and a basis $f_1,\dots ,f_m$ of $F$ such that
$$ \langle e_i,e_j\rangle =\langle f_i,f_j\rangle =0$$
and
$$ \langle e_i,f_j\rangle =0 \,\text{if}\, i+j\ne  m+1\quad \text{and}\quad \langle e_i,f_{m+1-i}\rangle =(-1)^{m-i}.$$
Finally $X$ is  such that $Xe_i=e_{i-1},\,\, Xf_i=f_{i-1}$.

Let $\xi $ be the matrix of the restriction of $X$ to $E$ or to $F$. Write an element $A\in {\mathfrak  g}$ as $2\times2$ matrix $A=(a_{i,j})$. Then
$$[X,A]=\begin{pmatrix}[\xi ,a_{1,1}]&[\xi ,a_{1,2}]\cr [\xi ,a_{2,1}]& [\xi ,a_{2,2}]\end{pmatrix}.$$
 Suppose that $v\in Q(X)$ and let
$$ v=e+f\,\,\, \text{ with }\,\,\,e=\sum x_ie_i,\,\,\,f=\sum y_if_i.$$
We get
$$ X\varphi   _v+\varphi   _vX=\begin{pmatrix}\xi \varphi   _{f,e}+\varphi   _{f,e}\xi &\xi \varphi   _{f,f}+\varphi   _{f,f}\xi \cr\xi \varphi   _{e,e}+\varphi   _{e,e}\xi &\xi \varphi   _{e,f}+\varphi   _{e,f}\xi \end{pmatrix}$$
where, for example $\varphi   _{e,e}$ is the map $f'\mapsto \langle f',e\rangle e$ from $F$ into $E$. Thus, for some $A$,
$$\xi \varphi   _{e,e}+ \varphi   _{e,e}\xi =\xi a_{2,1}-a_{2,1}\xi$$
In this formula, using the basis $(e_i),\, (f_i)$ replace all the
maps by their matrices.

 Then, as before, we have $\tr (\varphi   _{e,e}\xi ^q)=0$ for $1\leq q\leq   m-1$. If $e'=\sum x_if_i$ (the $x_i$ are the coordinates of $e$),
 then $\tr (\xi^q\varphi   _{e,e})$ is $\langle\xi ^qe,e'\rangle $. Thus, as in the other cases, we have $x_j=0$ for $j>m/2$ if $m$ is even and $j>(m+1)/2$ if $m$ is odd. The same thing is true for the $y_i$.

 If $m=2p$ is even, let $V_1=\oplus _{i\leq   p}({\mathbb  F}e_i\oplus {\mathbb
F}f_i)$ and $V_2=\oplus _{i>p}({\mathbb  F}e_i\oplus {\mathbb
F}f_i)$; write $v=v_1+v_2$ the corresponding decomposition of an
arbitrary element of $V$. Let $\delta _2$ be the Dirac measure at
the origin in $V_2$ and $dv_1$ a Haar measure on $V_1$. Then, as
in the unitary case, using the Fourier transform, we see that the
distribution $T$ must be a multiple of $dv_1\otimes \delta_2$.

 The endomorphism $a$ of $V$ defined by $a(e_i)=(-1)^ie_i$ and $a(f_i)=(-1)^{i+1}f_i$ belongs to $G$ and $aXa^{-1}=-X$. The map $u: (Y,v)\mapsto (-Y,-v)$
 belongs to $\widetilde G\setminus G$ so that the product $a\times u$ in $\widetilde G$ belongs to $\widetilde C\setminus C$. It clearly leaves $T$ invariant so that $T=0$.

 Finally if $m=2p+1$ is odd we put $V_1=\oplus _{i\leq   p}({\mathbb  F}e_i\oplus {\mathbb  F}f_i)$, $V_0={\mathbb  F}e_{p+1}\oplus {\mathbb  F}f_{p+1}$,
  $V_2=\oplus _{i\geq p+2}({\mathbb  F}e_i\oplus {\mathbb  F}f_i)$.
 As in the unitary case we find that $T=dv_1\otimes R\otimes \delta  _2$ with $R$ a distribution on $V_0$. As $-\Id\in C$ we see that $R$ must be even.
 Then again, define $a\in G$ by $a(e_i)=(-1)^ie_i$ and $a(f_i)=(-1)^if_i$ and consider $a \times u$ with $u(Y,v)=(-Y,-v)$.
 As before $a \times u\in \widetilde C\setminus C$ and leaves $T$ invariant so we have to take $T=0$.
 \hfill$\Box$\hfill

\def\cprime{$'$} \def\Dbar{\leavevmode\lower.6ex\hbox to 0pt{\hskip-.23ex
  \accent"16\hss}D} \def\cftil#1{\ifmmode\setbox7\hbox{$\accent"5E#1$}\else
  \setbox7\hbox{\accent"5E#1}\penalty 10000\relax\fi\raise 1\ht7
  \hbox{\lower1.15ex\hbox to 1\wd7{\hss\accent"7E\hss}}\penalty 10000
  \hskip-1\wd7\penalty 10000\box7}
  \def\polhk#1{\setbox0=\hbox{#1}{\ooalign{\hidewidth
  \lower1.5ex\hbox{`}\hidewidth\crcr\unhbox0}}} \def\dbar{\leavevmode\hbox to
  0pt{\hskip.2ex \accent"16\hss}d}
  \def\cfac#1{\ifmmode\setbox7\hbox{$\accent"5E#1$}\else
  \setbox7\hbox{\accent"5E#1}\penalty 10000\relax\fi\raise 1\ht7
  \hbox{\lower1.15ex\hbox to 1\wd7{\hss\accent"13\hss}}\penalty 10000
  \hskip-1\wd7\penalty 10000\box7}
  \def\ocirc#1{\ifmmode\setbox0=\hbox{$#1$}\dimen0=\ht0 \advance\dimen0
  by1pt\rlap{\hbox to\wd0{\hss\raise\dimen0
  \hbox{\hskip.2em$\scriptscriptstyle\circ$}\hss}}#1\else {\accent"17 #1}\fi}
  \def\bud{$''$} \def\cfudot#1{\ifmmode\setbox7\hbox{$\accent"5E#1$}\else
  \setbox7\hbox{\accent"5E#1}\penalty 10000\relax\fi\raise 1\ht7
  \hbox{\raise.1ex\hbox to 1\wd7{\hss.\hss}}\penalty 10000 \hskip-1\wd7\penalty
  10000\box7} \def\lfhook#1{\setbox0=\hbox{#1}{\ooalign{\hidewidth
  \lower1.5ex\hbox{'}\hidewidth\crcr\unhbox0}}}
\providecommand{\bysame}{\leavevmode\hbox
to3em{\hrulefill}\thinspace}
\providecommand{\MR}{\relax\ifhmode\unskip\space\fi MR }
% \MRhref is called by the amsart/book/proc definition of \MR.
\providecommand{\MRhref}[2]{%
  \href{http://www.ams.org/mathscinet-getitem?mr=#1}{#2}
} \providecommand{\href}[2]{#2}



\end{document}
